Photovoltaic Energy Harvesting for Millimeter-Scale Systems

by

Eunseong Moon

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy (Electrical and Computer Engineering) in the University of Michigan 2020

Doctoral Committee:

Professor Jamie D. Phillips, Chair Professor Pallab K. Bhattacharya Professor David Blaauw Associate Professor Cynthia Chestek Eunseong Moon

esmoon@umich.edu

ORCID iD: 0000-0003-3378-7930

© Eunseong Moon 2020

To my family

For their love and support

Acknowledgements

First, I would like to thank my advisor Professor Jamie Phillips for giving me the opportunity to work as a graduate student at the University of Michigan and the continuous support throughout my Ph.D. study. Over the past six years, he inspired me to think about the idea for research works and guided me to become an independent researcher. It was great time for me to work and study with him. I am sincerely thankful for his encouragement, patience, and mentorship.

I would like to express my gratitude to committee members, Professor David Blaauw, Professor Pallab K. Bhattacharya, and Professor Cynthia Chestek for their valuable feedback and suggestions regarding my research. Without their devotion and collaboration, I could not have finished my dissertation and Ph.D. works.

I would like to appreciate my mentor Dr. Alan Teran for teaching me the scientific background and fabrication processes of photovoltaic cells. Also, I thank the previous and current group members, Dr. Justin Easley, Minhyung Ahn, Michael Barrow, Martin Scherr, Armando Gil, Hannah Masten, Rebecca Lentz, and Sarah Puzycki for their friendship and cooperation.

I would like to thank my friends, Soomin Kim, Dr. Jihang Lee, Dr. Jihun Choi, Dr. Hyeongseok Kim, Dr. Youngbae Son, Dr. Kanghwan Kim, Wonjin Choi, Jiseok Gim, Jaeho Lim, Jaesung Jo, Jihun Lim, Charin Jang, Junghoon Kim, and Hojun Son. I enjoyed my time with you in Ann Arbor. Especially, I would like to appreciate Dr. Sunmin Jang. He supported and encouraged me from the beginning to the end of my Ph.D. study. Being a friend with Dr. Sunmin Jang is one of the great moments in my life. I would like to thank my collaborators and colleagues, Prof. Taekwang Jang, Prof. Inhee Lee, Dr. Gyouho Kim, Dr. Wootaek Lim, Jongyup Lim, Hyochan An, Dr. Anthony Aiello, Dr. Arnab Hazari, and Dr. Aniruddha Bhattacharya. Also, I would like to appreciate LNF staff members for teaching me and helping me to develop the process and recipes.

I would like to thank my mentors, Prof. Gyehong Park and Prof. Kangwook Kim. They encouraged and counselled me to study abroad for pursing my graduate study.

Finally, I would like to thank my family members Heesang Moon, Kyunghee Park, and Aehyun Moon for their endless love, support, and sacrifice. They are my strength and a real support for my life.

Table of Contents

Dedication	ii
Acknowledgements	iii
List of Figures	vi
List of Tables	xvii
Abstract	xviii
Chapter 1 Introduction	1
Chapter 2 Small-area Silicon Photovoltaic Cells	10
Chapter 3 Small-area GaAs Photovoltaic Cells	28
Chapter 4 Subcutaneous Phovoltaic Energy Harvesting	49
Chapter 5 Monolithic GaAs Photovoltaic Modules	66
Chapter 6 Vertical dual junction GaAs Photovoltaics	83
Chapter 7 Monolithically Integrated Microscale LED and Photovoltaic Module	99
Chapter 8 Conclusion and Futuer Works	115
Appendix	128
Bibliography	143

List of Figures

Figure

1.1.	IoT ² applications for biomedical, security, micro-robots and industrial monitoring (adopted	
	from [1])1	
1.2.	The standard terrestrial AM 1.5 solar spectrum	
1.3.	An equivalent circuit of the photovoltaic cell with parasitic resistances4	
1.4.	An example current and power versus voltage curves of the photovoltaic cell, illustrating	
	the short circuit current, the open circuit voltage and the maximum power point5	
1.5.	White and near infrared LED spectra compared with the solar spectrum7	
1.6.	Comparison chart of illuminance from indoor to outdoor conditions (adopted [21])7	
1.7.	Conceptual illustration of subcutaneous photovoltaic energy harvesting through tissue8	
2.1.	Schematic illustration of conventional high-efficiency solar cell (adopted from [29])11	
2.2.	(a) Schematic illustration of device structure and (b) optical microscope image of a	
	fabricated 1 mm ² device12	
2.3.	Simulated power conversion efficiency under 100 nW/mm^2 illumination at 800 nm	
	(dashed) and 850 nm (solid) for variable (a) base thickness and (b) base doping	
	concentration13	
2.4.	Simulated current-voltage characteristics of silicon photovoltaic cells with / without lightly	
	doped emitter under 660 nW/mm ² illumination at 850 nm14	

2.5.	Optical microscope images of the resulting DRIE isolation trenches15
2.6.	Measured current versus voltage characteristic of photovoltaic cell with 1 mm ² under dark
	condition18
2.7.	(a) Extracted J0 and (b) diode ideality factors versus P/A (mm ⁻¹) ratio from 1.4 mm ⁻¹ to
	35.84 mm ⁻¹ corresponding device size from 10 mm ² to 0.02 mm ² 18
2.8.	Measured (solid) and simulated (dashed) current-voltage characteristics of photovoltaic
	cell under 660 nW/mm ² illumination at 850 nm19
2.9.	Measured and simulated external quantum efficiency (EQE) characteristics for the 100 nm
	LPCVD Si_3N_4 passivated cell along with a surface reflectance curve (dashed) of cell
	between 375 nm and 1100 nm wavelength20
2.10.	Measured current versus voltage characteristics of different passivation layers for 10 mm ²
	cell under 850nm LED illumination with 660nW/mm ² 21
2.11.	Measured <i>J</i> - <i>V</i> under dark conditions for samples with different passivation layers21
2.12.	Measured J-V characteristics and corresponding conversion efficiency and comparison to
	device simulation with no surface and sidewall losses and an ideal Shockley-Queisser
	model
2.13.	Measured power conversion efficiency versus NIR illumination for varying device
	passivation and comparison to commercial c-Si (IXYS Corporation: KXOB22-12X1,
	[42])25
2.14.	Measured fill factor versus NIR illumination for varying device passivation and
	comparison to commercial c-Si (IXYS Corporation: KXOB22-12X1, [42])25
2.15.	An equivalent circuit model of PV diode with parasitic series and shunt resistances26

2.16.	Measured power conversion efficiency versus NIR illumination for varying device
	passivation and comparison to commercial c-Si. Simulated values using a diode model in
	(a) are shown assuming J0 and n values shown in the inset and varying shunt resistance in
	$(\Omega-cm^2)$
3.1.	Schematic diagram of conventional GaAs photovoltaic device structure
3.2.	(a) Schematic diagram of optimized device structure, (b) Optical microscope image of
	fabricated GaAs PV cell at mm-scale
3.3.	Simulated (a) energy band diagram and (b) profile of photogeneration rate under
	illumination
3.4.	Simulated maximum power density under 100 nW/mm ² illumination at 800 nm and 850
	nm versus variable base (a) doping concentration and (b) thickness
3.5.	Measured current density versus voltage curves of different passivation studies of 6.4 mm ²
	cell under (a) 850 nm NIR-LED illumination with 660 nW/mm^2 and (b) white LED
	illumination with 580 lux (1.38 μ W/mm ²)34
3.6.	Measured and Simulated (dashed) EQE characteristics of $(NH4)_2S + Si_3N_4$ passivated
	sample along with the wavelength range between 400 nm and 900 nm35
3.7.	Measured power conversion efficiency values of various passivation studies versus P/A
	(mm^{-1}) ratio from 1.69 mm ⁻¹ to 55.56 mm ⁻¹ corresponding device size from 6.4 mm ² to
	0.0052 mm ²
3.8.	Measured current versus voltage of 0.25 mm ² cell under dark conditions37
3.9.	Extracted reverse saturation current values of different passivation studies versus P/A
	(mm^{-1}) ratio from 1.69 mm ⁻¹ to 125 mm ⁻¹ corresponding device size from 6.4 mm ² to 0.001
	mm ²

3.17.	Measured efficiency versus variable NIR LED illumination conditions, compared to
	commercial c-Si cell (IXYS Corporation: KXOB22-12X1, [32]) and our custom silicon
	PV cell (discussed in Chapter 2) with well-optimized LPCVD Si_3N_4 passivation48
4.1.	The light propagation and the optical properties of biological tissues (adopted from
	[79,81])
4.2.	Optical properties of biological tissues, illustrating the optical transparency windows in
	NIR wavelength ranges (adopted from [20])
4.3.	Measured temperature of post-craniotomy subdural feline brain tissue under 850 nm NIR
	irradiation over 10 min exposure
4.4.	Current density versus voltage curves of silicon and GaAs photovoltaic cells under 1.06
	μ W/mm ² at 850 nm wavelength and 25 °C55
4.5.	External quantum efficiency spectra of silicon and GaAs photovoltaic cells
4.6.	Current density versus voltage curves of silicon and GaAs photovoltaic cells and
	comparison to Shockley–Queisser model under $1.06\mu W/mm^2$ at 850 nm wavelength57
4.7.	Temperature dependence of power density versus voltage ranging from room temperature
	(25 °C) to conventional body temperature (37 °C) for (a) Silicon and (b) GaAs
4.8.	Measured transmittance versus thickness of porcine skin and chicken breast under 1.08
	μ W/mm ² at 850 nm wavelength
4.9.	Power conversion efficiency versus thickness of porcine skin and chicken breast for (a)
	Silicon and (b) GaAs photovoltaic cells under 1.08 μ W/mm ² at 850 nm wavelength60
4.10.	Output power versus input power plots for (a) Silicon and (b) GaAs photovoltaic cells
	through 5 mm porcine skin and 10 mm chicken breast along with minimum 50 nW/mm^2

- 5.2. Schematic diagrams of (a) device structure illustrating PV cell junction, junction barrier isolation, and shunt leakage path, and (b) equivalent circuit model of the PV module......68

- 5.5. Measured *J-V* curves for varying number of PV cell series connection and (inset) corresponding power conversion efficiency at the maximum power point......72

5.9.	Comparison of measured P - V characteristics between PV arrays and single PV cell (a)
	under 850 nm NIR illumination at 1.02 $\mu W/mm^2$ and (b) under white LED illumination at
	586 lux
5.10.	Light intensity dependence of (a) power conversion efficiency and (b) fill factor for single
	PV and PV modules with p-GaAs and p-AlGaAs barrier junction isolation under NIR
	illumination78
5.11.	Light intensity dependence of (a) power conversion efficiency and (b) fill factor for single
	PV and PV modules with p-GaAs and p-AlGaAs barrier junction isolation under white
	LED illumination
5.12.	Optical microscope images of a wireless mm-scale sensor system with integrated PV
	module (a) before and (b) after encapsulation
5.13.	Monitored battery voltage output of this mm-scale system during the charging process
	under 110 lux indoor illumination
6.1.	Schematic illustration of (a) monolithically grown and (b) mechanically stacked multi-
	junction device structures (adopted from [104,106])85
6.2.	Schematic diagrams of (a) single dual-junction PV cell and (b) dual-junction PV module
	illustrating PV cell junction, junction barrier isolation, and shunt leakage path85
6.3.	Simulated current voltage characteristics of dual junction cells with variable thickness of a
	top cell under 850 nm NIR illumination at 1 μ W/mm ²
6.4.	Simulated external quantum efficiency (EQE) spectra for different thickness of top PV
	cells in the dual-junction structure with the fixed thickness of bottom PV cell at $3.5 \mu m \dots 87$
6.5.	Comparison of simulated <i>J-V</i> characteristics between single-junction and dual-junction PV
	cells under 850 nm NIR illumination at 1 μ W/mm ² 88

Optical microscope image of fabricated dual junction PV cell at µm-scale
Measured J-V and P-V characteristics of fabricated dual junction PV cells with 6.4 mm^2
size under 2.02 μ W/mm ² NIR illumination at wavelength of 850 nm90
Measured J-V and P-V characteristics of fabricated dual junction PV cells with 150 μ m x
150 μ m size under 2.02 μ W/mm ² NIR illumination at wavelength of 850 nm91
Optical microscope images of two different fabricated DJ PV modules that have 4 and 8
cells connected in series
Measured <i>J-V</i> and <i>P-V</i> characteristics of PV modules with 4 cells in series
Measured J-V and P-V characteristics of PV modules with 8 cells in series
Comparison of measured J-V characteristics between DJ and SJ PV cells under 850 nm
NIR illumination at 6.62 µW/mm ² 94
Measured EQE spectra of DJ and SJ PV cells and comparison with simulated EQE
spectrum of DJ PV cell
Comparison of measured J-V characteristics between SJ (8 cells in series) and DJ (4 cells
in series) PV arrays under 850 nm NIR illumination at $2.02 \mu W/mm^296$
Comparison of measured J-V characteristics between SJ (8 cells in series) and DJ (8 cells
in series) PV arrays under 850 nm NIR illumination at $2.02 \mu W/mm^296$
Conceptual illustration of a full system for recording and transmitting the neural signals
and a PV / LED module in the system for NIR wireless power transfer and data
communications (Drawn by Jongyup Lim)100
Experiment set up for measuring the NIR optical transmittance through the dura samples
using laser and LED illumination sources102

7.3.	Measured NIR transmittance through post-craniotomy feline dura samples under 850 nm
	laser illumination102
7.4.	Measured NIR transmittance through post-craniotomy NHP dura samples under 850 nm,
	970 nm and 1050 nm NIR LED illumination103
7.5.	Schematic diagram of monolithically integrated μ -LED / PV module illustrating tandem
	junction of PVs and junction barrier isolation between LED and PV104
7.6.	Optical microscope image of fabricated LED / PV module at μ m-scale104
7.7.	Measured I-V and P-V characteristics of dual junction PV cell under 850 nm NIR LED
	illumination at 100 $\mu W/mm^2105$
7.8.	Monte-Carlo simulation of NIR emission from a μ -LED through dura to the SPAD
	repeater, with estimated efficiency of 0.1% (photons received at SPAD with 100 μm
	diameter relative to photons emitted from LED) (Simulated by Michael Barrow)106
7.9.	Measured <i>I-V</i> characteristic of fabricated LED with 50 μ m x 50 μ m size107
7.10.	Measured photoluminescence (PL) versus wavelength curve of fabricated LED with 50 μm
	x 50 µm size
7.11.	Measured light output versus applied current curve of fabricated LED with 50 μm x 50 μm
	size108
7.12.	Detected SPAD output (10 µs deadtime) versus LED input pulses (40 µs pulse width, 100
	μs period, 1.54 applied voltage) (Collaborating with Jongyup Lim)109
7.13.	Measured <i>I-V</i> characteristics of PV cell before / after the shared connection to LED cathode
	under 850 nm NIR LED illumination at 150 μ W/mm ² 110
7.14.	Measured leakage current between LED cathode and PV anode with / without the isolation
	trench111

7.15.	Measured I-V characteristics of PV cell with / without the isolation trench under 850 nm
	NIR LED illumination at $150 \mu W/mm^2$ 111
7.16.	Extracted EQE versus applied current curves of variable size LEDs112
7.17.	(a) Measured temperature dependent photoluminescence spectra of InGaAs LED and (b)
	extracted peak intensity values versus variable temperature from 300 K to 13 K
	(collaborating with Anthony Aiello)113
8.1.	Cross-sectional concept diagram of the integration of GaAs PV / LED modules, silicon
	chips and neural probe116
8.2.	Schematic diagrams for (a) the front side and (b) back side of GaAs PV / LED module for
	the vertical integration, illustrating the through wafer via, interconnects and back
	contacts
8.3.	(a) Microscope and (b) scanning electron microscopy (SEM) images of through wafer vias
	of GaAs, deeper than 50 µm117
8.4.	(a) An image of the thinned GaAs wafer mounted to the flexible tape and (b) a microscope
	image of the backside of thinned wafer, exposing the TWVs and predefined alignment
	mark118
8.5.	(a) An optical microscope image of backside of thinned GaAs with TWVs, and (b)
	electrical characteristics measured through two GaAs TWVs in series (collaborating with
	Michael Barrow)118
8.6.	Photon detection probability along with wavelengths at different excess bias at room
	temperature (adopted from [122])120
8.7.	Optimized device parameters of DH AlGaAs LED with the emission wavelength at 780
	nm120

- 8.9. Simulated EQE spectra for variable aluminum mole fraction of top PV cells in the DJ structure with the fixed thickness of top PV cell at 400 nm and comparison with the normalized spectrum of commercialized white LED with 3000 K color temperature.....123

List of Tables

Table

1.1.	Energy harvesting sources for wireless power transfer (adopted from [7])2
2.1.	Optimized device parameters under 850 nm NIR illumination at 100 nW/mm ² 14
2.2.	Extracted diode parameters of 10 mm ² cells using 1 diode equation (1)22
2.3.	Device size dependence on device parameters of LPCVD passivated cells24
3.1.	Extracted diode parameters of 6.4 mm ² cells from 1-Diode equation
3.2.	Extracted diode parameters from a 2-Diode equation40
4.1.	Optimized device parameters of silicon and GaAs PV cells for NIR wavelength54
4.2.	Parameters measured at various locations on the mouse sample
6.1.	Optimized device parameters of dual junction PV cell and module, designed for NIR
	wavelength at 850 nm
6.2.	Comparison of measured device performances between DJ and SJ PV cells under 850 nm
	NIR illumination at 6.62 µW/mm ² 94
6.3.	Comparison of measured device performances between SJ and DJ PV modules with 4
	and 8 cells in series under 850 nm NIR illumination at 6.62 $\mu W/mm^297$

Abstract

The Internet of Things (IoT) based on mm-scale sensors is a transformational technology that can be applied to a wide range of applications for biomedical devices, surveillance, micro-robots and industrial monitoring. Energy harvesting approaches to power IoT have traditionally utilized thermal energy, mechanical vibrations and radio frequency electromagnetic radiation. However, the achievement of efficient energy scavenging for IoT at the mm-scale has been elusive. Here I show that photovoltaic cells at the mm-scale can be an alternative means of wireless power transfer to mm-scale sensors for IoT, utilizing ambient indoor lighting or intentional irradiation of near-infrared (NIR) LED sources through biological tissues, where the power conversion efficiency is limited by sidewall / perimeter recombination losses and shunt resistance.

First, I show that silicon photovoltaic cells at the mm-scale can achieve a power conversion efficiency of more than 17 % under 850 nm NIR irradiation at 1 μ W/mm² using the optimized sidewall passivation studies using LPCVD Si₃N₄ without the concern of shunt leakage degradation. However, fundamental material characteristics of silicon such as higher dark current limit the open circuit voltage and corresponding power conversion efficiency.

On the other hand, GaAs based PV cells have a possibility to boost the energy harvesting efficiency under low-flux conditions due to superior optical properties, range of tunable bandgap energies, low dark current, a large shunt resistance in comparison to crystalline silicon photovoltaic cells. GaAs photovoltaic cells at the mm-scale can achieve a power conversion efficiency of more than 30 % under 850 nm NIR irradiation at 1 μ W/mm² and around 20 % under white LED

illumination at 580 lux, which guarantees perpetual operation of mm-scale sensors.

The practical requirements of subcutaneous photovoltaic energy harvesting in terms of the optical properties of biological tissues, near-infrared transmittance of various tissue samples, the maximum exposure limit, and the body temperature effect are explored. Our results demonstrate that external infrared energy harvesting from ambient sources or intentional irradiation are sufficient to power mm-scale sensor systems utilizing silicon or GaAs PV cells that are specifically designed and optimized for energy harvesting in the NIR transparency window for biological tissue. Sufficient power generation is achieved for perpetual operation of mm-scale systems for implant depth of at least 15mm including hair / skin / muscle / bone under NIR illumination at 850 nm.

Furthermore, monolithic GaAs photovoltaic modules offer an efficient means for energy harvesting and direct battery charging in mm-scale systems. I demonstrated GaAs PV modules at the mm-scale with high efficiency under low-flux conditions, where AlGaAs junction barrier isolation provided a critical step in limiting shunt leakage current between series connected cells. We observed power conversion efficiency of 26.3 % under 850 nm infrared LED illumination at $1.02 \,\mu$ W/mm2 and 16.3 % under white LED illumination at 586 lux (1.4 μ W/mm2), with a 90 % voltage up-conversion efficiency to reach an operating voltage of 5 V for direct battery charging.

Dual-junction GaAs photovoltaic cells and modules at sub millimeter scale are demonstrated for efficient wireless power transfer for Internet of Things (IoT) and bio-implantable applications under monochromatic illumination. The dual-junction approach meets demanding requirements for these applications by increasing the output voltage per cell with reduced area losses from isolation and interconnects. A single PV cell (150 μ m x 150 μ m) based on the dual-junction design demonstrates power conversion efficiency above 22 % with greater than 1.2 V

output voltage under low-flux 850 nm near-infrared LED illumination at 6.62 μ W/mm², which is sufficient for batteryless operation of miniaturized CMOS IC chips. The output voltage of dual junction PV modules with 8 series-connected cells demonstrates greater than 10 V for direct battery charging while maintaining a module power conversion efficiency of more than 18 %.

Finally, I demonstrate monolithic PV / LED modules at the μ m-scale for brain-machine interfaces, enabling two-way power and data transfer in a through-tissue configuration. The dual junction GaAs PV cell provides sufficient power (~ 1 μ W) and voltage (> 1.4 V) for battery-less operation of CMOS circuitry under 850 nm NIR illumination at 100 μ W/mm² within the tissue exposure limit. The monolithically integrated InGaAs μ -LED emits NIR light pulses at detectable power levels (> 1 nW) under realistic energy harvesting levels from PV cells, where we have demonstrated example digital pulse detection with a single photon avalanche photodiode (SPAD). The wafer-level assembly plan for the 3D vertical integration of GaAs LED / PV modules, CMOS silicon chips, and neural probes is proposed, using the through-wafer via, wafer thinning, flip chip bonding, laser dicing, and final packaging.

CHAPTER 1

Introduction

1.1. Internet of Things (IoT)

The Internet of Things (IoT) is a transformative technology that can be applied for a variety of applications [1-2], which interconnects individual objects and devices to a network without human interfaces. Furthermore, continued scaling of electronic systems, and the proliferation of wireless mm-scale sensor networks have enabled a paradigm shift referred to Internet of Tiny Things (IoT²) [1-2]. These unique features of IoT² with small form factors at mm-scale open new capabilities of various applications for biomedical, security, micro-robots and industrial monitoring as shown in Fig. 1.1.



Figure 1.1. IoT^2 applications for biomedical, security, micro-robots and industrial monitoring (adopted from [1]).

1.2. Wireless power transfer for the IoT²

The IoT² necessitates a means of energy harvesting from ambient and stable sources to achieve self-powered devices with small form factors at mm-scale. Several different energy sources utilizing thermal energy [3], mechanical vibrations [4] and radio-frequency (RF) electromagnetic radiation [5-6] have been evaluated and tested as shown in Table 1.1, though miniaturization and reliability / stability of the ambient sources are still primary limiting factors. Photovoltaic energy harvesting provides an alternative means of wireless power transfer to these miniaturized low-power systems using ambient or intentional illumination sources, where the power requirement of mm-scale sensors for the energy-autonomous operation is roughly 10 μ W (active) and 0.5 nW (standby), respectively [8-11].

Power Source	Power Density
Photovoltaics (outdoors)	$15,000 \ \mu W/cm^2$
Photovoltaics (indoors)	$10 \mu\text{W/cm}^2$
Thermoelectric (5 °C gradient)	$40 \ \mu W/cm^2$
Piezoelectric (shoe inserts)	$330 \mu\text{W/cm}^3$
Vibration (building)	$300 \mu\text{W/cm}^3$
Ambient radio frequency	$< 1 \ \mu W/cm^2$

 Table 1.1. Energy harvesting sources for wireless power transfer (adopted from [7]).

1.3. Basic characteristics of photovoltaic cells

1.3.1 Photovoltaic effect

The photovoltaic effect refers to the conversion process of light energy into electrical energy. Light from sun or other sources is electromagnetic radiation or photons whose energy is

determined by the wavelength of light from the equation (1)

$$E = \frac{hc}{\lambda} \tag{1}$$

where *h* is Planck's constant, *c* is the speed of light and λ is the wavelength. For example, the sun emits light with a wide range of wavelengths from ultraviolet (UV) to infrared (IR) ranges. The spectrum reaching to the Earth is absorbed and scattered through the atmosphere and its intensity is dependent on the angle of incident. The standard spectrum is called Air Mass (AM) 1.5 [12] as shown in Fig. 1.2 when the sun is at the elevation angle of 42°, which has the integrated intensity at 1 mW/mm².



Figure 1.2. The standard terrestrial AM 1.5 solar spectrum.

1.3.2. Basic operation of photovoltaic cells

Photovoltaic cells usually based on p-n junction diodes absorb the incident photons above the bandgap and generate the electron-hole pairs that diffuse toward the junction. The generated carriers are separated and collected to the external circuit, generating the electrical power. The generated photocurrent is described by following equation (2)

$$J_{SC} = q \int b_s(E) QE(E) dE \tag{2}$$

where b_s is the incident spectral photon flux density, *QE* is the quantum efficiency. QE is a key parameter for the operation of photovoltaic cells, relevant to the efficiency of absorption, separation, and collection of carriers. Under the dark condition, photovoltaic cells behave like a normal rectifying diode in opposite direction to the photocurrent following the equation (3)

$$J_{dark}(V) = J_0(e^{qV/kT} - 1)$$
(3)

where V is the applied voltage, J_0 is the reverse saturation current, k is Boltzmann's constant and T is temperature. The equivalent circuit of the photovoltaic diode is illustrated in Fig 1.3.



Figure 1.3. An equivalent circuit of the photovoltaic cell with parasitic resistances.

The total current of photovoltaic cells from the superposition approximation as is described by following equation (4).

$$J(V) = J_{SC} - J_{dark}(V) = J_{SC} - J_0(e^{qV/kT} - 1)$$
(4)

For the open-circuited diode (J = 0), the maximum potential difference called the open circuit voltage (V_{OC}) is obtained, which is given by the equation (5).

$$V_{OC} = \frac{kT}{q} \ln\left(\frac{J_{SC}}{J_0} + 1\right) \tag{5}$$

The output power is given by the equation (6)

$$P = JV \tag{6}$$

where *P* has the its maximum (P_m) at the maximum power point. Another important parameter is the fill factor, describes the squareness of the *J*-*V* curve using the equation (7).

$$FF = \frac{P_m}{J_{Sc}V_{OC}} \tag{7}$$

The power conversion efficiency (η_c) of the photovoltaic cell is defined by the equation

$$\eta_C = \frac{P_m}{P_{in}} \tag{8}$$

where P_{in} is the incident power density.

(8)



Figure 1.4. An example current and power versus voltage curves of the photovoltaic cell, illustrating the short circuit current, the open circuit voltage and the maximum power point.

1.3.3. Parasitic resistances

In the actual photovoltaic cell, parasitic series and shunt resistances as show in Fig. 1.3 can degrade cell performance. The parasitic resistance components for a photovoltaic cell are described by the equation (9)

$$J = J_{SC} - J_0 \left[\exp\left(\frac{q(V+JAR_s)}{nkT}\right) - 1 \right] - \frac{V+JAR_s}{AR_{sh}}$$
(9)

where J_{SC} is the short-circuit current density, and R_S and R_{SH} are series and shunt resistance, respectively. The series resistance is related to the resistance of the cell material and contacts, governing the device operation at high current densities under high-flux outdoor or concentrated conditions. The shunt resistance relevant to the leakage of current through the cell or sides becomes increasingly important under low-flux conditions where short-circuit current values are small.

1.4. Photovoltaic energy harvesting for the IoT²

Photovoltaics are well known for efficient large scale power generation and for their use in self-powered electronic devices at the macroscale. While the physical dimensions of PV devices and systems may be reduced, miniaturization to the mm-scale present new challenges in achieving high conversion efficiency. Furthermore, sources such as ambient indoor lighting or infrared radiation for wireless power transfer differ dramatically from solar irradiance in terms of both spectral content and flux [13-16] as shown in Fig. 1.3, 1.4 and 1.5. Dark current and shunt conductive paths in photovoltaic cells become much more important at small dimensions and low flux ambient indoor or subcutaneous conditions [13-18] in comparison to typical outdoor solar irradiation (approximately a factor of 1,000 lower flux than AM 1.5). High-efficiency PV cells can meet the power requirements (> 50 nW/mm²) [8,13] of these systems through optimization of the spectral response in appropriate spectral windows: 425-650 nm [19] for ambient indoor lighting



and 700-1100 nm for the infrared transparency window for biological tissue [19-20].

Figure 1.5. White and near infrared LED spectra compared with the solar spectrum.



Figure 1.6. Comparison chart of illuminance from indoor to outdoor conditions (adopted [21]).

Previous works on photovoltaic wireless power transfer (WPT) for IoT^2 were cm-scale or larger and tested under extremely high intensity of laser illumination in mW range [22], comparable to the intensity used for laser therapy treatments [23]. Here we demonstrate that photovoltaic cells at mm-scale or below can provide power densities needed for the perpetual operation of miniaturized systems via low-level irradiation under indoor lighting or NIR



illumination at a wavelength of 850 nm in a through-tissue configuration.

Figure 1.7. Conceptual illustration of subcutaneous photovoltaic energy harvesting through tissue.

1.5. Thesis organization

This thesis includes an in-depth study of silicon and GaAs based photovoltaic energy harvesting for mm-scale systems under low-flux indoor or subcutaneous conditions from the theoretical device simulation / modeling to the actual device fabrication / packaging.

Chapter 2 introduces silicon photovoltaic cells at the mm-scale and their performance limiting factors in terms of the sidewall recombination loss and shunt resistance for the operation of miniaturized photovoltaic cells under low-flux conditions. The effective solutions to overcome these limiting factors are evaluated, using the optimized device structures and sidewall passivation studies.

Chapter 3 provides an in-depth study of GaAs based photovoltaic cells at the mm-scale to boost the energy harvesting efficiency under low-flux conditions. The optimized chemical and dielectric passivation studies on the sidewall and surface of GaAs and their effects on device performances are explored.

Chapter 4 demonstrates the feasibility of subcutaneous energy harvesting using mm-scale silicon and GaAs photovoltaic cells through the near-infrared optical transparency window of various tissues samples.

Chapter 5 shows an approach of monolithic GaAs photovoltaic modules, offering an efficient means for energy harvesting and direct battery charging in mm-scale systems without the voltage up-conversion loss. The AlGaAs junction barrier isolation scheme is introduced as a critical step in limiting shunt leakage current between series connected cells.

Chapter 6 explores vertically stacked dual-junction PV cell and module in the same GaAs material for direct powering to IoT and bio-implantable applications with narrowband spectral response and miniaturized device size at sub mm-scale. The effectiveness of dual junction approach to increase the output voltage per cell with reduced area losses from isolation and interconnects is discussed for the ultimate battery-less operation of miniaturized CMOS IC chips.

Chapter 7 introduces an approach of a monolithic GaAs based PV / LED module and a full system assembly at the μ m-scale for brain-machine interfaces such as "neural dust", enabling two-way optical power and data transfer in a through-tissue configuration.

Chapter 8 proposes future works for the wafer-level integration of neural probe assemblies including GaAs PV / LED wafer, CMOS silicon chips and neural probes through the through-wafer via, wafer thinning, and flip chip bonding. The modified and improved μ -LED design optimized for CMOS SPAD arrays are also proposed.

CHAPTER 2

Small-area Silicon Photovoltaic Cells

2.1. Introduction

Silicon photovoltaics is an attractive option for wireless energy harvesting in mm-scale systems that can be widely used for novel applications including IoT and bio-implants due to their excellent visible and NIR response, compatibility with silicon CMOS [24], and reduced cost, utilizing ambient sources from solar irradiation and indoor lighting or an intentional LED illumination. Conventional high efficiency silicon photovoltaic cells at cm-scale or larger as shown in Fig 2.1 are well established for solar energy and optimized for visible and NIR wavelength range with above 90 % external quantum efficiency (EQE), where low flux solar response has demonstrated a power conversion efficiency of 13.5 % under 1 μ W/mm² [25] and 13.1 % under 3 μ W/mm² AM 1.5 illumination [26]. However, specific requirements for device operation under extremely low-flux conditions, miniaturization of device size down to mm-scale, and a stackable device configuration to integrate silicon PV cells with existing miniaturized CMOS systems have not been addressed, where perimeter/sidewall recombination [13, 21] and shunt resistance [21] are expected to be critical. In this chapter, mm-scale silicon photovoltaic cells are explored for low-flux energy harvesting.

2.2. Device structure & Fabrication

The device structure of conventional high-efficiency silicon PV cells [27-29] with above 25 % efficiency under AM 1.5 solar irradiation [12] is fabricated by using a high-quality crystalline silicon wafer with a few hundred microns thick ($300 - 500 \mu m$) for efficient light absorption as shown in Fig. 2.1. The front surface of wafer is textured into inverted pyramids using wet or dry etching processes to reduce the reflection losses, which is passivated with high-quality silicon oxide for lowering the surface recombination. The differential heavy doping of n layer near the front metal fingers and rear point contacts on the back side of wafer for heavily doped p-type region are used to reduce the additional front and rear surface recombination losses.



Figure 2.1. Schematic illustration of conventional high-efficiency solar cell (adopted from [29]).

Our silicon photovoltaic cells were designed based on the structure of commercial silicon PV cells [27-29]. The baseline device structure consists of a thick p-type silicon base layer on heavily doped p-type silicon substrate, diffused lightly doped n-type emitter near the surface between heavily doped emitter contacts on the top surface with the goal of improving carrier collection and passivated front surface using high-quality oxide layer, as shown in Fig. 2.2 (a). In contrast to conventional solar silicon PV cells, the baseline geometry utilizing a heavily doped substrate and lightly doped base is selected in this work to provide top contacts to facilitate a stacked configuration for mm-scale systems [8] and guarantee controllable thickness and doping concentration for optimization of the device structure to specific spectral ranges and characterization of shunt resistance of the device. Furthermore, anti-reflection coating layers were used rather than the surface texturing because the recombination losses from surface and sidewall are critical for mm-scale systems under low-flux applications.



Figure 2.2. (a) Schematic illustration of device structure and (b) optical microscope image of a fabricated 1 mm² device.

The detailed device parameters are simulated using Sentaurus Device [30], using built-in values for silicon material parameters at 300 K and neglecting the edge effects including surface recombination and sidewall recombination. The base thickness and doping concentration were optimized for irradiation between 800 nm and 850 nm at a power density of 100 nW/mm². The dependence of power conversion efficiency on base layer thickness and doping concentration are

shown in Fig. 2.3. Base thickness near 35 μ m, corresponding to the optical absorption depth of silicon [31] at 300 K, and base doping concentration near $2x10^{17}$ cm⁻³ were found to provide near optimal performance under the NIR illumination conditions simulated.



Figure 2.3. Simulated power conversion efficiency under 100 nW/mm² illumination at 800 nm (dashed) and 850 nm (solid) for variable (a) base thickness and (b) base doping concentration.

The inclusion of a surface n-type emitter with thickness of 340 nm and concentration of 2.5×10^{16} cm⁻³ was found to significantly increase the short-circuit current and corresponding power conversion efficiency due to the improved carrier collection throughout the device, as shown in Fig. 2.4. Optimized device parameters are summarized in Table 2.1.



Figure 2.4. Simulated current–voltage characteristics of silicon photovoltaic cells with / without lightly doped emitter under 660 nW/mm² illumination at 850 nm.

Parameter	Value	Unit
Base thickness	35	μm
Base doping concentration	2 x 10 ¹⁷	cm ⁻³
N-type surface layer thickness	0.34	μm
N-type surface layer doping	2.5 x 10 ¹⁶	cm ⁻³
Anti-reflection Si ₃ N ₄ layer	100	nm
Emitter width	3	μm
Emitter depth	0.46	μm

Table 2.1. Optimized device parameters under 850 nm NIR illumination at 100 nW/mm².

A lightly-doped emitter with 1.2 μ m thickness and 5x10¹⁸ cm⁻³ peak concentration was formed through a subsequent 300 nm reactive ion etch of the top surface. The main huddle during device fabrication is closely connected to deep trenches (> 40 μ m) for bottom p-contacts and device isolation. The well-established deep reactive-ion etching (DRIE) was used to etch the epitaxial silicon layer down to the heavily doped substrate as shown in Fig. 2.5. The patterning process over the deep trench was not easy due to poor sidewall coverage of spin coated photoresist (PR) and unwanted PR residues in the deep trenches, which was solved by using a sticky PR named SPR 7.0, lowering PR spinning speed below 1000 rpm, and using a low temperature annealing process around 85 °C. Aluminum contacts were fabricated for n-type and p-type layers using conventional photolithography, electron beam evaporation, and wet chemical etching processes. The detailed device fabrication recipe is included in Appendix. Due to the strong sidewall / perimeter dependence of small area PV cells [13,21], variable device area was studied in the range of 0.02 mm² to 10 mm², corresponding to a perimeter/area (P/A) ratio of from 35.84 mm⁻¹ to 1.4 mm⁻¹.



Figure 2.5. Optical microscope images of the resulting DRIE isolation trenches.
2.3. Surface / Sidewall passivation studies

Several sidewall and surface passivation layer processes were investigated, since surface recombination is expected to have a major impact on device performance for small mm-scale devices operating under low-flux conditions. Passivation layers studied in this work include low-pressure chemical vapor deposition (LPCVD) of Si₃N₄, LPCVD a-Si due to good coverage on the deep exposed sidewall of LPCVD process [32] and resulting improved passivation for dangling bonds of the surface and sidewall [33-34], SiO₂ via dry thermal oxidation due to reduction in interface states by the high quality material growth directly using the existing silicon material on the sidewall and surface, plasma enhanced chemical vapor deposition (PECVD) of Si₃N₄, PECVD SiO₂ due to lower growth temperature of PECVD process below 400 °C and compatibility to other possible chemical passivation studies [35-40], and no passivation for comparison. Further details on the six different passivation layers under study are as follows:

- 1) 50 nm LPCVD Si_3N_4 at 800 °C + 50 nm PECVD Si_3N_4 at 380 °C.
- 2) 25 nm LPCVD a-Si at 560 °C + 100 nm PECVD Si₃N₄ at 380 °C.
- 3) 40 nm thermally grown dry SiO₂ at 900 °C + 50 nm LPCVD Si₃N₄ at 800 °C
 - + 100 nm PECVD SiO₂ at 380 °C.
- 4) 100 nm PECVD Si₃N₄ at 380 °C.
- 5) 100 nm PECVD SiO₂ at 380 °C.
- 6) control sample without passivation.

2.4. Characterization methods

Electrical current density versus voltage (J-V) characteristics under dark and illumination were measured using Keithley 2400 / 4200 semiconductor characterization systems. Low-flux illumination conditions were controlled by a microscope-compatible 850 nm infrared light emitting diode and calibrated power-meter. Power conversion efficiency was measured for irradiation at a power density of 660 nW/mm², which is extremely dim in comparison to the power density of AM 1.5 sunlight of 1000 μ W/mm² [12]. The EQE spectrum was measured on select samples using a system equipped with a halogen white light source, lock-in amplifier, monochromator and calibrated photodetector.

2.5. Results

2.5.1. Sidewall recombination loss

Dark current measurements for variable size of PV cells that are passivated by thermal silicon dioxide as shown in Fig. 2.6 were done to demonstrate the size dependence on device characteristics and corresponding numerical parameters for dark current were obtained by fitting the forward biased region from 0 V to 0.4 V to the diode equation

$$J = J_0 \left[\exp\left(\frac{qV}{nkT}\right) - 1 \right]$$
(1)

where *J* is the total current density, *V* is the applied voltage, *n* is the diode ideality factor, *k* is the Boltzmann constant, *T* is the temperature and J_0 is the reverse saturation current. Extracted diode parameters as shown in Fig. 2.7 show a strong perimeter dependence, ranging from 0.443 to 15.9 nA/mm² for J_0 and from 1.285 to 1.723 for the ideality factor. This unwanted increase in diode parameters including reverse saturation current and ideality factor is attributed to an increase in Shockley–Read–Hall (SRH) recombination in the space charge region near the sidewall. In comparison to simulation results as shown in Fig. 2.8, measured values of V_{OC} and corresponding efficiency are still substantially lower than simulated values assuming no losses from surface and sidewall recombination and surface reflection.



Figure 2.6. Measured current versus voltage characteristic of photovoltaic cell with 1 mm² under dark condition.



Figure 2.7. (a) Extracted J_0 and (b) diode ideality factors versus P/A (mm⁻¹) ratio from 1.4 mm⁻¹ to 35.84 mm⁻¹ corresponding device size from 10 mm² to 0.02 mm².

The primary source of efficiency reduction is likely due to non-ideal thermal silicon dioxide passivation of the p-type sidewall resulting in reduced minority carrier diffusion length, requiring improved passivation of the sidewall to approach the simulated efficiency values. Improved passivation of the sidewall may be achieved by using a-Si or Si₃N₄ [33] to approach the simulated efficiency values. Losses associated with the high doping concentration at the emitter surface interface, such as free carrier absorption or Auger recombination are other efficiency limiting factors. An optimized doping profile of the surface n-type emitter through selective etching between the light harvesting and metal finger regions are required, as well as improved design of the surface passivation layer to serve as an anti-reflection coating and layer to effectively reduce bulk and sidewall recombination losses.



Figure 2.8. Measured (solid) and simulated (dashed) current-voltage characteristics of photovoltaic cell under 660 nW/mm² illumination at 850 nm.

2.5.2. Passivation studies

Silicon cells with variable passivation layers and optimized surface n-type emitter were fabricated. The passivation layers served the dual purpose of sidewall passivation and an anti-reflection coating, where the surface reflectance was optimized at a wavelength of 800 nm. The 100 nm Si₃N₄ is expected to provide 1-2 % surface reflectance at 850 nm as shown in Fig. 2.9.



Figure 2.9. Measured and simulated external quantum efficiency (EQE) characteristics for the 100 nm LPCVD Si_3N_4 passivated cell along with a surface reflectance curve (dashed) of cell between 375 nm and 1100 nm wavelength.

Fabricated device sizes were 1 mm² and 10 mm². A comparison of *J*-*V* results is shown in Fig. 2.10, where LPCVD Si₃N₄ and LPCVD a-Si passivation demonstrate the highest power conversion efficiency of 17.12 % and 16.16 %, respectively. The improved efficiency originates primarily from an increase in V_{OC} , with relatively similar J_{SC} values. From the EQE measurement as shown in Fig 2.10, the device structures including the base thickness and anti-reflection layer are well optimized both for visible and NIR wavelength with above 80% EQE over this wavelength range. Dark *J*-*V* characteristics are shown in Fig. 2.11, with results from parameter extraction summarized in Table 2.2. The LPCVD Si₃N₄ and a-Si passivation processes demonstrate a clear reduction in reverse saturation current and reduction in ideality factor from 1.968 to 1.393, demonstrating that the LPCVD processes is effectively passivating the deep sidewall with Si₃N₄ and a-Si. The improved dark J-V characteristics demonstrate a clear agreement with the measured improvements in V_{OC} for the Si₃N₄ and a-Si processes.



Figure 2.10. Measured current versus voltage characteristics of different passivation layers for 10 mm² cell under 850nm LED illumination with 660nW/mm².



Figure 2.11. Measured *J-V* under dark conditions for samples with different passivation layers.

Passivation	J_0 (A/mm ²)	п
LPCVD Si ₃ N ₄	5.31 x 10 ⁻¹²	1.393
LPCVD a-Si	8.87 x 10 ⁻¹²	1.419
Dry SiO ₂	3.54 x 10 ⁻¹⁰	1.851
PECVD Si ₃ N ₄	7.03 x 10 ⁻¹¹	1.498
PECVD SiO ₂	1.04 x 10 ⁻⁹	1.943
Unpassivated	4.07 x 10 ⁻¹⁰	1.968

Table 2.2. Extracted diode parameters of 10 mm² cells using equation (1).

2.6. Discussion

The maximum efficiency achieved in the cells for this study is 17.12% for the LPCVD Si_3N_4 under low-flux 850 nm LED illumination. This provides a power density of 113 nW/mm², above the desired value of 100 nW/mm² for mm-scale systems [8,21]. The results are also improved over previously reported commercial cm-scale c-Si photovoltaic cells that were tested under the low-flux solar spectrum condition of 3000 nW/mm², with reported efficiency of 13.1% [26].

The limitation on power conversion efficiency of these cells are mainly attributed to a reduction in V_{OC} , as illustrated in Fig. 2.12 comparing measured and calculated / simulated results. The calculated results from Shockley–Queisser (SQ) limit [41] assumed complete absorption of all photons above band gap of silicon, no non-radiative recombination losses and lossless transport of excited carriers. The simulated results from the drift-diffusion device simulation in Sentaurus Device used the practical material parameters of silicon and a 2D cross section of device, with no sidewall or surface recombination. The evident reduction in measured V_{OC} in comparison to values obtained from the SQ limit and device simulation arises from the increase in reverse saturation

current due to non-radiative recombination losses including sidewall and surface recombination as indicated in equation (2)

$$V_{OC} = \frac{nkT}{q} ln \left(\frac{J_{SC}}{J_0} + 1\right) \tag{2}$$

where J_{SC} is the short-circuit current, J_0 is the reverse saturation current, n is the diode ideality factor, k is the Boltzmann constant and T is the temperature.



Figure 2.12. Measured *J-V* characteristics and corresponding conversion efficiency and comparison to device simulation with no surface and sidewall losses and an ideal Shockley–Queisser model.

As shown in Table 2.3, V_{OC} values and corresponding conversion efficiency decreases for the smaller 1 mm2 devices, suggesting that sidewall recombination is more important for mmscale applications in comparison to conventional cm-scale photovoltaic cells. Degradation in conversion efficiency under low incident light intensity by shunt resistance compared to the negligible impacts of series resistance can also impact the utility of PV cells for energy harvesting applications.

Passivation	LPCVD Si ₃ N ₄		LPCVD a-Si	
Size (mm ²)	10	1	10	1
P/A (mm ⁻¹)	1.18	3.13	1.18	3.13
J_{SC} (nA/mm ²)	388.29	403.18	387.47	404.66
$V_{OC}\left(\mathrm{V} ight)$	0.409	0.384	0.394	0.359
Efficiency (%)	17.12	15.91	16.16	15.53
$J_0 (\mathrm{A/mm^2})$	5.31x10 ⁻¹²	2.36x10 ⁻¹¹	8.87x10 ⁻¹²	1.45x10 ⁻¹¹
Ideality factor	1.393	1.502	1.419	1.356
Fill factor	0.711	0.678	0.699	0.705

Table 2.3. Device size dependence on device parameters of LPCVD passivated cells under illumination of 660 nW/mm^2 at a wavelength of 850 nm.

Power conversion efficiency and fill factor under variable intensity illumination were examined for the cells with LPCVD Si₃N₄ and a-Si passivation, as shown in Fig. 2.13 and Fig. 2.14. The measured devices exhibit a decrease in efficiency with reduced illumination, with similar behavior for both passivation techniques. Measured results for a commercial c-Si solar cell (IXYS Corporation: KXOB22-12X1) [42] with 22 % power conversion efficiency under AM 1.5 illumination are shown for comparison, which exhibits a more dramatic decrease in fill factor and corresponding efficiency with reduced illumination. To examine the possible influence of efficiency degradation due to shunt leakage, conversion efficiency as shown in Fig. 2.16 was simulated using a diode model (Fig. 2.15) with assuming extracted J_0 and n values from dark current measurement of tested cell and variable shunt resistance using an equation

$$J = J_{sc} - J_0 \left[exp\left(\frac{q(V+JAR_s)}{nkT}\right) - 1 \right] - \frac{V+JAR_s}{R_{sh}}$$
(2)

where J_{SC} is the short circuit current, J_0 is the reverse saturation current, n is the diode ideality factor, T is the temperature, A is the device area, k is the Boltzmann constant, R_S is the parasitic series resistance and R_{Sh} is the parasitic shunt resistance.



Figure 2.13. Measured power conversion efficiency versus NIR illumination for varying device passivation and comparison to commercial c-Si (IXYS Corporation: KXOB22-12X1, [42]).



Figure 2.14. Measured fill factor versus NIR illumination for varying device passivation and comparison to commercial c-Si (IXYS Corporation: KXOB22-12X1, [42]).

The LPCVD Si₃N₄ and a-Si passivated cells studied in this work demonstrate a shunt resistance above 10 M Ω -cm², sufficient to prevent degradation in efficiency for the range of illumination studied. Efficiency degradation with illumination follows expected behavior where cells are limited by the dark current (reverse saturation current density, *J*₀). Further improvements in cell efficiency will therefore require reduction in J₀, where techniques such as atomic layer deposition of Al₂O₃ [33-34] and chemical surface treatments including NH₄F [35-36], (NH₄)₂S [37-38] and H₂S [39-40] may be beneficial.



Figure 2.15. An equivalent circuit model of PV diode with parasitic series and shunt resistances.

2.7. Conclusion

Small area Si photovoltaic cells were optimized based on simulation results for energy harvesting applications in mm-scale systems. High EQE above 80% and power conversion efficiency exceeding 17 % are demonstrated under low-flux NIR illumination. The device performance was dramatically improved by LPCVD passivation. In contrast to conventional c-Si PV, the cells in this work demonstrate stable performance under low illumination intensity that is limited by dark current rather than shunt leakage. The good performance of the small-area silicon

cells under low illumination conditions are promising for through-tissue infrared energy harvesting and IoT applications, where further improvements may be achieved by additional measures to reduce recombination losses at interfaces.



Figure 2.16. Measured power conversion efficiency versus NIR illumination for varying device passivation and comparison to commercial c-Si. Simulated values using a diode model in (a) are shown assuming J_0 and n values shown in the inset and varying shunt resistance in (Ω -cm²).

CHAPTER 3

Small-area GaAs Photovoltaic Cells

3.1. Introduction

Photovoltaic energy harvesting provides an alternative means of wireless power transfer (WPT) for mm-scale systems [37,43-45], where the ambient solar irradiation and indoor lighting or an intentional LED illumination could provide sufficient energy to PV cells and conventional high efficiency PV cells [46] convert this wavelength region efficiently with above 90 % external quantum efficiency (EQE). The key challenges to obtaining highly efficient PV energy harvesting are that device performances of PV cells at mm-scale or smaller under low-flux illumination are degraded by shunt resistance [13,21,26] and sidewall / perimeter recombination losses [13,47]. Previous works on photovoltaic WPT [37,43-45] for low-flux applications are cm-scale or larger and tested under extremely high intensity of laser illumination in mW range [45,48], which is not suitable for fully remote device operation. One of reported silicon PV cells at mm-scale in the previous chapter addressed and relatively overcame these challenges, which had above 17 % power conversion efficiency under low-flux NIR illumination below 1 μ W/mm² at a wavelength of 850 nm. However, fundamental material characteristics of silicon such as higher dark current and shunt leakage compared to other III-V compound semiconductors limit the open circuit voltage and corresponding power conversion efficiency. On the other hand, GaAs based PV cells have a possibility to boost the energy harvesting efficiency due to superior optical properties matched to

the desired wavelength region, low dark current and low shunt leakage, which worked successfully under low-flux indoor conditions [13,21] and could provide sufficient power for mm-scale systems. In this work, the details of mm and sub-mm scale GaAs photovoltaics cells optimized for low-flux energy harvesting applications and their performance limiting factors are explored.

3.2. Device structure & Fabrication

The structure of conventional high-efficiency GaAs PV cells [46] as shown in Fig. 3.1 consists of a thicker base (several µm thick) and thinner emitter (several hundred nm thick) layers with addition of higher bandgap window and back surface field layers. The cells are based on n-p or p-n designs because minority diffusion lengths can be greater than the absorption depth for either doping types. The higher bandgap window and back surface field layers using lattice matched InGaP or AlGaAs materials are incorporated to reduce the surface recombination losses by reflecting minority carriers away from the surface.



Figure 3.1. Schematic diagram of conventional high-efficiency GaAs photovoltaic device structure.

Our device structure grown by molecular beam epitaxy (MBE) utilizing the structure of conventional high efficiency GaAs solar cells consists of n-base and p-emitter layers with addition of p-Al_{0.8}Ga_{0.2}As window and n-Al_{0.3}Ga_{0.7}As back surface field layers on a semi-insulating GaAs substrate for the possible PV array design as shown in Fig. 3.2 (a). Devices were fabricated using the chlorine based reactive-ion etching (RIE) to etch the epitaxial layers down to the n+ contact layer, the conventional photolithography using photoresists and physical vapor deposition (PVD) of thin metal films (Detailed fabrication recipe is included in Appendix)



Figure 3.2. (a) Schematic diagram of optimized device structure, (b) Optical microscope image of fabricated GaAs PV cell at mm-scale.

The layer structure of GaAs photovoltaic (PV) cell for low-flux energy harvesting was optimized based on simulation results using Synopsys Sentaurus [30] using widely accepted material parameters at 300 K and neglecting recombination effects from the surface/sidewall. The simulated energy band diagram and the profile of photo-generation rate under NIR illumination were indicated in Fig. 3.3. The device parameters are designed both for visible and NIR wavelength ranges under low intensity illumination at 100 nW/mm², where the device performance is

dominated by shunt resistance. The base doping concentration of 10^{17} cm⁻³ and base thickness of 2.75 µm result in the maximum output power density under 100 nW/mm² at a wavelength range between 800 nm and 850 nm as shown in Fig. 3.4. The optimized device parameters are summarized in Fig. 3.2 (a). Devices with varying area were fabricated to study the influence of sidewall / perimeter recombination losses [13,21,26], ranging from 0.001 mm² to 6.4 mm², corresponding to perimeter / area (P/A) ratio from 125 mm⁻¹ to 1.7 mm⁻¹.



(b)

Figure 3.3. Simulated (a) energy band diagram and (b) profile of photogeneration rate under illumination.



Figure 3.4. Simulated maximum power density under 100 nW/mm² illumination at 800 nm and 850 nm versus variable base (a) doping concentration and (b) thickness.

3.3. Passivation studies

Various passivation processes were studied including dielectric passivation and chemical passivation methods to reduce the sidewall/perimeter recombination losses, including plasma

enhanced chemical vapor deposition (PECVD) Si₃N₄, (NH4)₂S [49-51] and NH₄OH [52] with subsequent PECVD Si₃N₄, NH₄OH with subsequent atomic layer deposition (ALD) Al₂O₃ [52]. The dielectric passivation layers using PECVD Si₃N₄ and ALD Al₂O₃ were used to reduce the interface traps and passivate the dangling bonds on the surface and sidewall. The chemical passivation studies using (NH4)₂S and NH₄OH were used to etch the native oxide and passivate the surface and sidewall using the additional sulfur bonds in (NH4)₂S or make the hydroxylated surface using NH₄OH ready for dielectric passivation deposition [52]. Further details of the different passivation processes are as follows:

1) 100 nm PECVD Si₃N₄ at 380 °C.

2) (NH₄)₂S (23% in H₂O) for 10 min at room temperature + 100 nm PECVD Si₃N₄ at 380 °C
3) NH₄OH (29 % in H₂O) for 3 min at room temperature + 100 nm PECVD Si₃N₄ at 380 °C,
4) NH₄OH (29 % in H₂O) for 3 min at room temperature + 25 nm ALD Al₂O₃ at 150 °C,
5) a sample without passivation.

3.4. Results

3.4.1. Passivation layer

The *J-V* characteristics of 6.4 mm² GaAs PV cells passivated with the five different passivation layers under 660 nW/mm² illumination at a wavelength of 850 nm and 580 lux (1.38 μ W/mm²) white LED illumination are shown in Fig. 3.5. A 100 nm PECVD Si₃N₄ passivation layer matched to both visible and NIR wavelength ranges as shown in Fig. 3.5 increased the short circuit current (*J_{SC}*) compared to control sample without passivation by reducing the surface reflectance and the surface recombination loss. The ammonium sulfide ((NH₄)₂S) treatment encapsulated with PECVD Si₃N₄ layer for long-term stability of layers [53] improved both the short

circuit current density (J_{SC}) and the open circuit voltage (V_{OC}), which results in the maximum power conversion efficiency above 30 % under 850 nm NIR illumination with 660 nW/mm² and around 20 % under white LED illumination with 580 lux can supply sufficient power to a mm-scale system with the minimum power requirement of 50 nW/mm² [8,13].



Figure 3.5. Measured current density versus voltage curves of different passivation studies of 6.4 mm² cell under (a) 850 nm NIR-LED illumination with 660 nW/mm² and (b) white LED illumination with 580 lux (1.38 μ W/mm²).

The EQE measurement as shown in Fig. 3.6 on this (NH4)₂S passivated sample shows that the device structure is well optimized for visible and NIR wavelength ranges with above 70 % EQE and 5-7 % discrepancy between simulated and measured EQE values is observed over the desired wavelength region due to the surface reflection from metal fingers. Further studies of different passivation layer effects on device performance for variable size of cells from 0.0052 mm² to 6.4 mm² corresponding perimeter/area (P/A) ratio 55.56 mm⁻¹ to 1.69 mm⁻¹ are shown in Fig. 3.7. The device performance of cells with varying size cells is strongly dependent on the P/A ratio, illustrating the critical impact of sidewall recombination loss as one of the most critical limiting factors for these small-area PV cells [13,47], as discussed further in discussion section.



Figure 3.6. Measured and Simulated (dashed) EQE characteristics of $(NH_4)_2S + Si_3N_4$ passivated sample along with the wavelength range between 400 nm and 900 nm.



Figure 3.7. Measured power conversion efficiency values of various passivation studies versus $P/A (mm^{-1})$ ratio from 1.69 mm⁻¹ to 55.56 mm⁻¹ corresponding device size from 6.4 mm² to 0.0052 mm².

The $(NH_4)_2S$ treatment improved V_{OC} values of various size PV cells and this V_{OC} improvement is closely relevant to the reduction in the reverse saturation current (J_0) as indicated in equation (1)

$$V_{OC} = \frac{nkT}{q} \ln\left(\frac{J_{SC}}{J_0} + 1\right) \tag{1}$$

where *n* is the ideality factor, *k* is the Boltzmann constant, *T* is the temperature, J_0 is the reverse saturation current density, and J_{SC} is the short circuit current density. The (NH₄)₂S treatment etched the native oxide [49-51] on the surface/sidewall and effectively passivated the dangling bonds with additional sulfur bonds, resulting in the reduction of J_0 . The NH₄OH treatment reduced J_{SC} due to increased optical scattering from the rough surface though the NH₄OH treatment etched the native oxide [52] to achieve more electronically favorable surface properties before the dielectric passivation. The 25 nm ALD Al₂O₃ did not provide an optimal anti-reflection match to the desired wavelength range, which increased the surface reflectance over the desired wavelength region more than the optimized 100 nm PECVD Si_3N_4 layer, resulting in a reduction in J_{SC} . The detailed device parameters regarding p-n junction diode equations and recombination losses were extracted from results of dark current measurement.

3.4.2. Dark current measurements

Dark current measurements were conducted on all PV cells with varying P/A ratio to examine the impact of surface passivation properties. The dark *J*-*V* plots of 0.25 mm² samples were shown in Fig. 3.8 and the extracted J_0 parameters shown in Fig. 3.9 obtained by curve fitting to the forward bias region between 0 V and 0.7 V using the 1-diode equation (2)

$$J = J_0 \left[exp\left(\frac{qV}{nkT}\right) - 1 \right] \tag{2}$$

where J is the total current density, J_0 is the reverse saturation current, V is the applied voltage, k is the Boltzmann constant, T is the temperature.



Figure 3.8. Measured current versus voltage of 0.25 mm² cell under dark conditions.

The extracted J_0 parameters for all passivation studies shown in Fig. 3.9 show the clear P/A ratio dependence due to increased sidewall/perimeter recombination losses as expected. The dramatic reduction in J_0 over all measured PV cells from $(NH_4)_2S$ treatment was also observed in Fig. 3.9. For example, extracted J_0 of 6.4 mm² PV cells summarized in Table 3.1 was reduced from 0.9038 pA/mm² for the sample without passivation to 0.0673 pA/mm² for the sample passivated with $(NH_4)_2S$ + PECVD Si₃N₄, agreeing with the improvement in V_{OC} as indicated in Fig. 3.5.



Figure 3.9. Extracted reverse saturation current values of different passivation studies versus P/A (mm⁻¹) ratio from 1.69 mm⁻¹ to 125 mm⁻¹ corresponding device size from 6.4 mm² to 0.001 mm².

Passivation	$J_0 (\mathrm{PA/MM}^2)$	n
Si ₃ N ₄	0.6157	1.939
$(NH_4)_2S + Si_3N_4$	0.0673	1.801
$NH_4OH + Si_3N_4$	0.5853	1.933
$NH_4OH + Al_2O_3$	0.4532	1.906
Unpassivated	0.9038	1.969

Table 3.1. Extracted diode parameters of 6.4 mm² cells from 1-Diode equation.

The diode ideality factor (*n*) of 6.4 mm² cells summarized in Table 3.1 supported previous results that the ideality factor values near n = 2 for samples are dominated by the space charge region (SCR) recombination losses mainly connected to the sidewall recombination losses and the (NH₄)₂S treatment with subsequent Si₃N₄ deposition reduced the diode ideality factor from 1.969 to 1.801, suggesting that the sulfur treatment unpinned the Fermi-level [53] and reduced the SCR recombination losses from surface and sidewall regions [54-55]. Numerical parameters relevant to the perimeter recombination losses can be extracted by curve fitting to the forward bias region using the 2-diode equation (3)

$$J = J_{01} \left[exp\left(\frac{qV}{kT}\right) - 1 \right] + J_{02} \left[exp\left(\frac{qV}{2kT}\right) - 1 \right]$$
(3)

where J_{01} , J_{02} are the saturation current densities for carrier recombination in the quasi-neutral region and in the space charge region, respectively. The extracted J_{02} along with P/A ratio are plotted in Fig. 3.10.



Figure 3.10. Extracted J_{02} parameters along with P/A Extracted J_{02} parameters along with P/A (mm⁻¹) ratio from 1.69 mm⁻¹ to 125 mm⁻¹.

The extracted J_{02} exhibiting the strong linear P/A dependence [47,56] can be expanded as (4)

$$J_{02} = J_{02B} + J'_{02P} \left(\frac{P}{A}\right)$$
(4)

using the linear curve fitting to J_{02} using (4) are summarized in Table 3.2, which indicate that the effect of Si₃N₄ passivation mainly reduced the bulk recombination current density from 1.934 pA/mm² to 0.7831 pA/mm² rather than the perimeter recombination loss from 0.4882 pA/mm to 0.4596 pA/mm compared to the control sample without passivation. The additional NH₄OH treatment with Si₃N₄ passivation also had a slight improvement in the perimeter recombination loss and made the optimized passivation to the bulk recombination only. ALD Al₂O₃ passivation reduced the bulk recombination losses significantly from 1.934 pA/mm² to 0.0438 pA/mm², but had little impact on reducing perimeter recombination losses dramatically from 1.934 pA/mm² to 0.1409 pA/mm² for the bulk recombination and from 0.4882 pA/mm to 0.1714 pA/mm for the perimeter recombination, suggesting that the exposed mesa edges were effectively passivated with additional sulfur bonds.

Passivation	J_{02B}	J_{02P}	$S_0 L_S$
	(pA/mm^2)	(pA/mm)	(cm^2/s)
Si ₃ N ₄	0.7831	0.4596	13.66
$(NH_4)_2S+Si_3N_4$	0.1409	0.1714	5.09
$NH_4OH + Si_3N_4$	0.313	0.4507	13.4
$NH_4OH + Al_2O_3$	0.0438	0.4518	13.43
Unpassivated	1.934	0.4882	14.51

Table 3.2. Extracted diode parameters from a 2-Diode equation.

The important parameters relevant to J'_{02P} can be extracted using (5) [47,56]

$$J_{02P}' = q n_i S_0 L_S \tag{5}$$

where n_i is the intrinsic carrier concentration, S_0 is the surface recombination velocity and L_S is the effective surface diffusion length. The extracted S_0L_S product of the (NH₄)₂S treated sample was 5.09 cm²/s where the high efficiency GaAs based solar cells had $S_0L_S < 1$ cm²/s [47,56-57]. Further improvements in the device structure and passivation studies are still needed to overcome the perimeter recombination losses.

3.4.3. Shunt resistance

The degradation from the parasitic shunt resistance dominating the device operation under low-flux illumination was investigated for the sample with $(NH_4)_2S + Si_3N_4$ passivation varying the illumination intensities as shown in Fig. 3.11.



Figure 3.11. Measured (solid) power conversion efficiency values under NIR illumination at a wavelength of 850 nm for the $(NH_4)_2S + Si_3N_4$ passivated PV cell and for commercial c-Si solar cell [42] for comparison. Simulated (dashed) values varying shunt resistance in (Ω -cm²) are shown using extracted J₀ and n diode parameters shown in the inset.

The simulated values for the comparison were calculated using the diode equation (6)

$$J = J_{sc} - J_0 \left[exp\left(\frac{q(V+JAR_s)}{nkT}\right) - 1 \right] - \frac{V+JAR_s}{R_{sh}}$$
(6)

where J_{SC} is the short circuit current, R_S is the parasitic series resistance and R_{Sh} is the parasitic shunt resistance, using numerical extracted diode parameters from Table 3.1 and varying the shunt resistance values in Ω -cm². As shown in Fig. 3.11, the performances of measured GaAs PV cell under various NIR illumination conditions from 30 nW/mm² to 2.69 μ W/mm² were free of the shunt resistance degradation with above 10 M Ω -cm² shunt resistance value that is consistent with the previous GaAs PV cell results though the commercial crystalline silicon solar cell (IXYS Corporation: KXOB22-12X1) [42] with above 22 % power conversion efficiency under AM 1.5 solar irradiation shows the clear performance degradation especially in fill factor from the shunt resistance under dim visible and NIR illumination around 1 μ W/mm² as shown in Fig. 3.12.





Figure 3.12. Measured current density versus applied voltage characteristics with corresponding maximum power conversion efficiency and comparison to Shockley–Queisser model, device simulation with no surface and sidewall recombination losses and commercial silicon PV (IXYS Corporation: KXOB22-12X1, [42]) under (a) NIR illumination and (b) white LED illumination.

3.5. Discussion

The power conversion efficiency for a 6.4 mm² cell is greater than 30 % under 850 nm NIR illumination at 660 nW/mm² and around 20 % under white LED illumination at 580 lux (1.38 μ W/mm²) through the optimization of device structure and improvement in *V*_{oc} from optimized sidewall passivation studies using (NH₄)₂S and PECVD Si₃N₄. This efficiency approaches the expected value from simulation results using a 2D device schematic and well-established material parameters of GaAs with no surface and sidewall recombination losses, as shown in Fig. 3.12. The parasitic shunt resistance of optimized GaAs PV cell is on the order of 10 MΩ-cm² range as shown in Fig. 3.11, which can sufficiently overcome the shunt resistance degradation in fill factor under variable visible and NIR illumination conditions in compassion to commercial silicon PV cell that show a drastic fill factor degradation as shown in Fig. 3.13. Furthermore, our silicon PV cell that

was discussed in previous Chapter 2 maintained the similar fill factor values for GaAs under extremely dim light condition due to well-optimized passivation studies on the surface and sidewall using LPCVD Si_3N_4 [32] as shown in Fig. 3.13 (a), suggesting that silicon PV cells can also overcome the shunt resistance degradation using a proper passivation layer for low-flux device operating conditions.



Figure 3.13. Measured fill factor versus variable (a) NIR and (b) white LED illumination conditions, compared to commercial c-Si cell (IXYS Corporation: KXOB22-12X1, [42]) and our custom silicon PV cell (discussed in Chapter 2) with well-optimized LPCVD Si3N4 passivation.

The J_0 and J_{02} values extracted from dark J-V measurements are gradually increased with increasing P/A ratio, which shows a dramatic reduction in dark current from 1.934 to 0.1409 pA/mm² for the bulk recombination current and from 0.4882 to 0.1714 pA/mm for the perimeter recombination coefficient from the (NH₄)₂S treatment encapsulated with PECVD Si₃N₄, attributed to the sulfur passivation of surface states [49-51]. The measured V_{OC} values in Fig. 3.14 agree with theoretically calculated values using the extracted diode parameters in Table 3.2 with the fixed EQE of photogenerated current for variable size PV cells, showing a reduction from the increased reverse saturation current originated from sidewall/perimeter recombination losses. The deviation of measured V_{OC} for small PV cells below 200 µm sidewall length is relevant to J_{SC} degradation.



Figure 3.14. Comparison with calculated (dashed) and measured V_{OC} values for square GaAs cells with varying the sidewall length.

The measured J_{SC} values as shown in Fig. 3.15 exhibit a sharp drop as cell size decreases though theoretical J_{SC} should be independent of device size. One of the major J_{SC} limiting factors for small PV cells is optical shadowing from metal fingers, pads and interconnects in the dark harvesting area, suggesting a novel device design using vertical dual junction (discussed in Chapter 6) for extremely small PV cells at sub-mm scale.



Figure 3.15. Comparison with calculated (dashed) and measured J_{SC} values for square GaAs cells with varying the sidewall length, assuming the fixed EQE around 80 % for theoretical calculation.

The overall power conversion efficiency values are shown in Fig. 3.16, and are limited by factors including an unavoidable sidewall recombination loss, optical shadowing from metal pads and deal area near sidewall. Therefore, further improvements in power conversion efficiency approaching Shockley–Queisser (SQ) limit [41] assuming complete absorption of all photons above band gap, no non-radiative recombination losses and lossless collection of excited carriers indicated in Fig. 3.12 will require continued reduction in edge effects, where the surface and sidewall passivation using wide bandgap materials matched to the lattice constant of GaAs such

as InGaP [58] or AlGaAs [59] may be required. The surface reflectance from metal pads might be reduced from transparent metal anode contacts such as Indium Tin Oxide (ITO) [60], Aluminumdoped Zinc Oxide (AZO) [60] or Ti/ITO [58]. Overall, single GaAs PV cells showed superior device performances than silicon PV cells compared to custom (discussed in Chapter 2) and commercial silicon PV cells (IXYS Corporation: KXOB22-12X1) [32] under broad illumination conditions as shown in Fig. 3.17 due to higher output voltage originated from the material characteristics, which is promising for more sophisticated device structures such as PV array and multi-junction PV cells.



Figure 3.16. Comparison with calculated (dashed) and measured power conversion efficiency values for square GaAs cells with varying the sidewall length.



Figure 3.17. Measured efficiency versus variable NIR LED illumination conditions, compared to commercial c-Si cell (IXYS Corporation: KXOB22-12X1, [32]) and our custom silicon PV cell (discussed in Chapter 2) with well-optimized LPCVD Si3N4 passivation.

3.6. Conclusion

GaAs photovoltaic cells at mm-scale and μ m-scale were optimized for wireless energy harvesting applications for mm-scale systems. Power conversion efficiency values beyond 30 % under low-flux NIR illumination below 1 μ W/mm² and around 20 % under indoor lighting around 600 lux were achieved. The (NH₄)₂S and PECVD Si₃N₄ passivation methods improved the device performance dramatically by reducing the surface/sidewall recombination losses. The optimized device structure was free of performance degradation from shunt leakage under low-flux illumination. High efficiency GaAs photovoltaic cell under low-flux illumination makes wireless photovoltaic energy harvesting for mm-scale systems into a feasible approach, where further improvements may be accomplished by further reduction in edge effects and optical shadowing from electrical contacts.

CHAPTER 4

Subcutaneous Photovoltaic Energy Harvesting

4.1. Introduction

Wireless biomedical implantable devices are prospective technologies that can be applied to a variety of applications for monitoring physiological variables [61-64]. For these implantable applications, low-power systems on the mm-scale [8,65-68] with efficient energy harvesters from ambient and stable sources are essential to make these technologies practical. Several different energy sources utilizing thermal energy [3,69,70] and mechanical vibrations [4,71,72], and radiofrequency (RF) electromagnetic radiation [5,6,73-75] have been evaluated and tested, though miniaturization and reliability / stability of the ambient sources are still primary limiting factors. Wireless power transfer via RF inductive coupling [5,6,73-75] is currently used in implantable systems due to highly efficient power transfer around 58 % at 13.56 MHz through the tissue with 250 mm² implanted coil area [74]. However, the power transfer efficiency is highly dependent on the distance between primary and secondary coils [74-75], decreasing power transfer efficiency exponentially. Efficiency also decreases dramatically as implantable device size decreases to sub mm-scale and below due to lateral and angular misalignments [76] and weak coupling [75] with mm-scale antenna receivers. An ultrasound source can also safely send power similar to the RF source, having an issue with the miniaturization due to bulky ultrasound transducers. On the other hand, Biological tissue also provides a means of wireless power transfer in the near-infrared (NIR) spectral region, where there are two optical transparency windows in the 650 nm to 1350 nm range (First: 650 nm ~ 950 nm, Second: 1000 nm ~ 1350 nm) [20,77,78]. Photovoltaic cells can efficiently convert in this NIR spectral region [29,46] with external quantum efficiency (EQE) approaching 100 %, and are commonly utilized for high-efficiency solar cells. Photovoltaic cells for NIR subcutaneous energy harvesting face challenges in achieving high efficiency under low irradiance conditions in cells of small area, where shunt conductance [13,33] and perimeter recombination losses can dramatically degrade performance [13]. Such cells are far more sensitive to shunt and recombination losses in comparison to solar cells that are typically cm-scale or larger and operating under irradiance that is orders of magnitude higher. Previous work on NIR photovoltaic infrared energy harvesting for biomedical implants [44] utilized PV cells on the cmscale utilizing relatively high laser light irradiation (mW range), comparable to the intensity used for laser therapy treatments [23]. Here we demonstrate that our silicon and GaAs photovoltaic cells at mm-scale from previous chapters can meet practical requirements for subcutaneous energy harvesting and provide power densities needed for the perpetual operation of implantable devices via low-level irradiation at a wavelength of 850 nm in a through-tissue configuration.

4.2. Optical properties of biological tissues

The optical properties of biological tissues are important to estimate the light propagation, absorption and scattering through various tissue types as shown in Fig. 4.1. for the feasibility of subcutaneous energy harvesting and designing highly efficient PV cells. The optical properties of tissues [20,79,80] are determined in terms of the absorption coefficient, μ_a (cm⁻¹), the scattering coefficient μ_s (cm⁻¹), the scattering function $p(\theta, \psi)$ (sr⁻¹) where θ is the deflection angle of scatter and ψ is the azimuthal angle of scatter and the refractive index of the tissue n'. For thicker tissues where the orientations of scattering are random due to multiple scattering events, the dependent of ψ is averaged and ignored. The averaged θ through the multiple scattering is described by the anisotropy of scatter $g = \langle \cos \theta \rangle$. These optical coefficients are subject to variations in blood, water, collagen and fiber contents.



Figure 4.1. The light propagation and the optical properties of biological tissues (adopted from [79,81]).

The measured optical properties of tissues including oxygenated blood, deoxygenated blood, skin and fatty tissues in Fig. 4.2 showed that absorption and scattering from several tissues were lowest in the NIR regions between 650 nm and 1350 nm [20,79], which are called optical transparency windows of biological tissue (First: 650 nm ~ 950 nm, Second: 1000 nm ~ 1350 nm). The peak at the absorption between 950 nm and 1000 nm is from the water in the blood. At wavelengths longer than 1350 nm, the absorption by water and lipids is also increased. Silicon and GaAs photovoltaic cells can efficiently covert the NIR wavelength range that is well matched to the first transparency window of biological tissues.


Figure 4.2. Optical properties of biological tissues, illustrating the optical transparency windows in NIR wavelength ranges (adopted from [20]).

4.3. Maximum NIR exposure limit

The maximum NIR exposure limit connected to the corresponding tissue temperature increase due to NIR illumination is a key issue for PV energy harvesting under subcutaneous conditions, following the American National Standards Institute (ANSI) standards [82]. We performed experiments to compare our results to ANSI standards. We implanted a micro temperature probe [83] beneath the dura and measured the temperature rise of the surrounding tissue in response to NIR LED irradiation in a post-craniotomy ex vivo feline. The measured temperature was increased around 2.5 °C for input irradiation at 0.8 mW/mm² over a 20 minute NIR LED irradiation at 850 nm wavelength as shown in Fig. 4.3. This temperature increase is expected to be a worst-case scenario, since active blood flow in cerebral tissue and cerebrospinal

fluid will aid in maintaining temperature. Given this measurement, the maximum NIR exposure limit at 850 nm wavelength was around 0.64 mW/mm² (below the NIR limit of 1.36 mW/mm² using ocular extended sources from ANSI [82]) to achieve a temperature increase of ~2 °C which is the recommended temperature limit for biological tissues [84].



Figure 4.3. Measured temperature of post-craniotomy subdural feline brain tissue under 850 nm NIR irradiation over 10 min exposure.

4.4. Cell performance under NIR illumination

We have studied both silicon and GaAs PV cells in previous chapters, where silicon offers advantages of compatibility with microelectronics technology, while GaAs offers superior light absorption properties, low dark current, and high shunt resistance. The silicon and GaAs PV cell designs were optimized for NIR illumination conditions using device simulations. The optimized device parameters for silicon and GaAs PVs under low-flux NIR illumination between 800 nm and 850 nm are summarized in Table I.

Material	Silicon		GaAs		
Parameter	Туре	Value	Туре	Value	Unit
Base thickness	р	35	n	2.75	μm
Base doping	р	$2 \ge 10^{17}$	n	1017	cm ⁻³
Emitter thickness	n	0.34	р	0.5	μm
Emitter doping	n	2.5 x 10 ¹⁶	р	$4 \ge 10^{18}$	cm ⁻³
Anti-reflection Si ₃ N ₄ layer	-	100	-	100	nm

Table 4.1. Optimized device parameters of silicon and GaAs PV cells for NIR wavelength.

Fabricated cells utilized highly optimized surface passivation and anti-reflection layers to minimize perimeter recombination effects and the surface reflection at a wavelength of 800 nm; 50 nm low-pressure chemical vapor deposition (LPCVD) Si₃N₄ + 50 nm plasma enhanced chemical vapor deposition (PECVD) Si₃N₄ for Silicon and 100 nm PECVD Si₃N₄ with (NH₄)₂S surface treatment for GaAs. The baseline performance of the PV cells is shown in Fig. 4.4 under 1.06 μ W/mm² LED irradiance at a wavelength of 850 nm, which is tested under extremely dim irradiance conditions in comparison to AM 1.5 sunlight conditions of 1000 μ W/mm² and the ANSI maximum exposure limit around 1.36 mW/mm² and represents an approximate irradiance scenario for charging under subcutaneous conditions. The current-voltage characteristics are shown in Fig. 4.4, demonstrating power conversion efficiency values of 17.82 % for silicon PV and 31.63 % for GaAs PV. The short circuit current density (*J_{SC}*) is similar for both silicon and GaAs, indicating similar conversion of the infrared flux to photocurrent.



Figure 4.4. Current density versus voltage curves of silicon and GaAs photovoltaic cells under $1.06 \,\mu\text{W/mm}^2$ at 850 nm wavelength and 25 °C.

The external quantum efficiency (EQE) spectra shown in Fig. 4.5 confirms J_{SC} results, with above 80% EQE for both silicon and GaAs over the desired NIR range between 700 nm and 850 nm. The metal fingers used in the cell design are a primary factor limiting J_{SC} and EQE where approximately 7% of the light is reflected by metal coverage on the top surface. The primary difference in power conversion efficiency between silicon and GaAs cells is the variation in open circuit voltage (V_{OC}), which tracks the material bandgap energy. The performance of the silicon and GaAs photovoltaic cells are limited by non-radiative perimeter, surface, and Shockley-Read-Hall (SRH) recombination losses [13] in practical PV devices, reducing the V_{OC} below the theoretical Shockley–Queisser (SQ) limit [41] assuming complete absorption of all photons above band gap, no non-radiative recombination losses and lossless transport of excited carriers. As shown in Fig. 4.6, while these PV cells demonstrate high power conversion efficiency, the SQ limit is calculated to be 32% for silicon and 53% for GaAs under 660 nW/mm² at a wavelength of 850 nm.



Figure 4.5. External quantum efficiency spectra of silicon and GaAs photovoltaic cells.

4.5. Temperature dependence

The operating temperature of biomedical implantable devices should also be considered, where body temperature ranges between 36 °C and 37 °C, in contrast to the typical room temperature of 25 °C. The increase in operating temperature can degrade the device performance by increasing the thermal carrier generation and corresponding increase in reverse saturation current and decrease in open circuit voltage.



Figure 4.6. Current density versus voltage curves of silicon and GaAs photovoltaic cells and comparison to Shockley–Queisser model under $1.06 \,\mu$ W/mm² at 850 nm wavelength.

The temperature dependence of the power density versus voltage is shown in Fig. 4.7 (a) and (b) for silicon and GaAs cells, respectively, exhibiting a reduction in V_{OC} of 2.09 mV/°C for silicon and 2.23 mV/°C for GaAs. The corresponding reduction in the power conversion efficiency in this temperature range is 0.097 %/°C for silicon and 0.069 %/°C for GaAs. These values are consistent with theoretical temperature dependence of PV cells [86] where such minor variations in conversion efficiency can generally be neglected; i.e. room-temperature characteristics provide an adequate representation of energy harvesting performance.



Figure 4.7. Temperature dependence of power density versus voltage ranging from room temperature (25 °C) to conventional body temperature (37 °C) for (a) Silicon and (b) GaAs.

4.6. Subcutaneous energy harvesting

We tested the feasibility of subcutaneous photovoltaic energy harvesting with variable thickness of tissue models via porcine skin and chicken breast to approximate properties of human skin [86-90] and muscle [78]. Initially, an infrared LED at 850 nm was aligned to photovoltaic cells at a fixed distance and the incident illumination density adjusted by the applied voltage to the infrared LED was scanned using a calibrated photodetector. The tissue samples with variable thickness were placed between the LED and PV cell to measure current-voltage characteristics in a through-tissue configuration. The measured transmittance of tissue samples with variable thickness and the dependence of PV cell output power density on tissue thickness were shown in Fig. 4.8 and 4.9 for irradiation under $1.08 \,\mu$ W/mm² at a wavelength of 850 nm. The chicken breast model exhibits a near linear dependence of transmittance and power density versus thickness rather than an exponential dependence of absorption, which might be caused by the possible variation in thickness of samples.



Figure 4.8. Measured transmittance versus thickness of porcine skin and chicken breast under $1.08 \,\mu\text{W/mm}^2$ at 850 nm wavelength.



Figure 4.9. Power conversion efficiency versus thickness of porcine skin and chicken breast for (a) Silicon and (b) GaAs photovoltaic cells under $1.08 \,\mu W/mm^2$ at 850 nm wavelength.

Optical characteristics of chicken breast was roughly extracted by the Beer-Lambert relation [78]:

$$P_{out} = P_0 \ e^{-\alpha d} \tag{1}$$

where *Pout* is the electrical power density produced by the cell, P_0 is the incident infrared power density, α is the attenuation coefficient, and d is the tissue thickness. An extracted attenuation coefficient of the chicken breast sample is around 1.706 cm⁻¹ at wavelength of 850 nm. Optical attenuation will occur via absorption and scattering, depending on the cell structure [79], and portions of blood, chromophores and pigments in the tissue. The constant attenuation coefficient for the chicken breast samples suggests a homogeneous medium, providing a good model for optical penetration into uniform soft tissue samples. The attenuation coefficient is similar to prior reports for human skin of 0.37 ± 0.12 cm⁻¹ [77] and subcutaneous adipose tissue of 1.1 ± 0.03 cm⁻¹ ¹ over the wavelength range between 620 nm and 1000 nm, and tumor samples [88] with attenuation coefficients of 3.29 ± 1.02 cm⁻¹ and 4.77 ± 0.77 cm⁻¹ at 789 nm wavelength. The power density dependence for harvesting through porcine skin exhibits a sharp attenuation near surface, suggesting an inhomogeneous medium. Optical transmission through human skin occurs via three primary layers [77,78,89]: the epidermis (100 µm thick), dermis (1 - 4 mm thick) and subcutaneous fat (1 - 6 mm thick). Attenuation in the epidermis and dermis is dominated by Mie scattering [91] via collagen fibers, where attenuation is reduced for latter propagation in fatty tissue. The porcine skin model therefore represents a good approximation to transmission through skin with high density of collagen fibers [86] in the dermis compared to human skin, and represents a worst case scenario for IR attenuation. The dependence of output power density versus input irradiance is shown in Fig. 4.10 for silicon and GaAs samples of 5 mm porcine skin and 10 mm chicken breast. Above irradiance of approximately 100 nW/mm², the harvesting efficiency is approximately constant, corresponding to the linear relationship on the log-log scale of Fig. 4.10. The energy harvesting efficiency decreases below irradiance of 100 nW/mm², attributed to the regime where dark current density in the PV cells approaches the photo-generated current.



Figure 4.10. Output power versus input power plots for (a) Silicon and (b) GaAs photovoltaic cells through 5 mm porcine skin and 10 mm chicken breast along with minimum 50 nW/mm² operating power of low-power microelectronics and minimum $10 \,\mu$ W/mm² low-level light therapy (LLLT).

The approximate requirement to power mm-scale systems is 50 nW/mm², which is demonstrated for all tissue samples in Fig. 4.10 for irradiance above 2.3 μ W/mm² for silicon and 1.3 μ W/mm² for GaAs. This irradiance condition is within an acceptable range of operation, and is below the typical minimum power density of 10 μ W/mm² that is safely used in low-level laser therapy (LLLT) [23] for medical treatments and below the maximum NIR exposure limit with 1.36 mW/mm² from ANSI, and produces a slight rise (below 0.5 °C) in the temperature of tissue / PV with negligible temperature degradations for PVs [45].

We studied a more complex energy harvesting scenario using NIR transmission through a previously dissected mouse to include complex combinations of hair, skin, bone, muscle, and organs. The dissected mouse sample was placed between the LED and PV cell for seven specific sections of the mouse sample, as shown in Fig. 4.11. Energy harvesting was measured using LED irradiance from above, with PV cells placed beneath the mouse. PV performance for the seven locations are summarized in Table II.



Figure 4.11. Photo of dissected mouse and mounted PV cells used to measure NIR energy harvesting.

#	Thickness [mm]	Input <u>-</u> power [µW/mm ²]	GaAs		Silicon	
			Output power [µW/mm ²]	Efficiency [%]	Output power [µW/mm ²]	Efficiency [%]
Point 1	7	134	5.12	3.82	2.82	2.1
Point 2	10	134	2.74	2.05	1.48	1.1
Point 3	4	134	12.24	9.13	7.75	5.79
Point 4	6	134	8.04	6.00	4.85	3.62
Point 5	10	134	3.4	2.53	2.07	1.54
Point 6	12	134	1.19	0.89	0.82	0.61
Point 7	15	134	0.29	0.21	0.17	0.12

 Table 4.2.
 Parameters measured at various locations on the mouse sample.

LED irradiation at 134 μ W/mm² (within the maximum exposure limit ~ 640 μ W/mm²) demonstrated stable harvesting capabilities at all seven locations. The ability to demonstrate energy harvesting at these locations, particularly point 7, which is a 15mm thick thorax region with high tissue density, shows great promise for infrared power transfer. Our silicon and GaAs cells demonstrate the ability to power biologically implanted mm-scale systems under low NIR irradiance conditions (approximately 1 μ W/mm²). Beyond the power generated by the PV cells, the implantable system will require an interface to directly power the system or to charge a battery. We have previously demonstrated that energy harvesting circuitry can exceed 78 % [92] at similar scale and low-flux conditions using a series/parallel PV network to match the charging voltage required for a battery without the need for voltage up-conversion. Further improvements in the cell structure will require the encapsulation of photovoltaic cells with bio-compatible and transparent polymer packaging materials such as poly(methyl methacrylate) (PMMA) [93] and polydimethylsiloxane (PDMS) [44,94] or glass [95] for long-term stability to reduce the toxicity

concern of arsenic compounds.

4.7. Conclusion

We show that photovoltaic cells at the mm-scale can achieve a power conversion efficiency of more than 17 % for silicon and 31 % for GaAs under 1.06 μ W/mm² infrared irradiation at 850 nm, which is the extremely dim light condition compared to the maximum exposure limit around 1.36 mW/mm² from ANSI and the. These photovoltaic cells demonstrate highly efficient energy harvesting through various biological tissue samples from ambient sunlight, or irradiation from infrared sources such as used in present-day surveillance systems by utilizing the near infrared (NIR) transparency window between the 650 nm and 1350 nm wavelength range. Sufficient power generation above 50 nW/mm² is achieved for perpetual operation of mm-scale systems for implant depth of at least 15 mm including hair / skin / muscle / bone under 850 nm NIR illumination at 134 μ W/mm².

CHAPTER 5

Monolithic GaAs Photovoltaic Modules

5.1. Introduction

The most critical issue in ensuring perpetual operation of systems is the overall power generation. In contrast to large-area applications, cost per unit area is a secondary factor in mmscale systems due to the small PV area (and hence, low cost). This enables the use of highperformance materials such as III-V compound semiconductors. Furthermore, to maximize energy capacity, the battery of the mm-scale systems often has a high open-circuit voltage, which is several times higher than the typical open circuit voltage of a single photovoltaic cell. A switched capacitor network as shown in Fig. 5.1 (a) is one of the typically used methods to achieve voltage up-conversion, where switching and resistive losses limit efficiency to approximately 50 % [92]. The other prevalent method for voltage up-conversion is to use a large off-chip inductor as shown in Fig. 5.1 (b), interfering with the miniaturization and the integration to mm-scale systems. Direct series/parallel connections of PV cells provide an appealing alternative for voltage up-conversion, where a PV network with over 80 % power conversion efficiency has been demonstrated [92]. However, monolithic PV arrays present several challenges in minimizing losses associated to device isolation and shunt leakage paths between series connections [18,96]. Monolithic silicon PV arrays present several challenges including low voltage generation (large number of seriesconnected cells required to achieve desired voltage) and low optical absorption strength (thick

absorber regions are required, making device isolation problematic) [18,97]. The larger voltage generation and high optical absorption strength of GaAs and related compound semiconductors provide a much more attractive platform for monolithic PV modules.



Figure 5.1. Schematic diagrams of (a) a switched capacitor network and (b) a bulk off-chip inductor for the DC-DC voltage up-conversion (adopted from [92]).

Previously, a laser power converter based on a six-cell GaAs PV module array was demonstrated at the mm-scale with conversion efficiency greater than 52 % under monochromatic illumination at 13.2 W/cm² (132 mW/mm²), with efficiency limited by perimeter recombination and shunt leakage through the semi-insulating GaAs substrate [18]. In this work, we present monolithic GaAs-based PV modules at the mm-scale operating under low flux conditions (< 10

 μ W/mm²) as a means to power IoT systems or bio-implantable sensors without the requirement for DC-DC voltage up-conversion.



Figure 5.2. Schematic diagrams of (a) device structure illustrating PV cell junction, junction barrier isolation, and shunt leakage path, and (b) equivalent circuit model of the PV module.

5.2. Experiment

We used a baseline PV cell structure (Fig. 3.1 (a)) grown by molecular beam epitaxy (MBE) based on our previously reported high-efficiency single-junction GaAs PV cells, where the critical limiting factor from exposed sidewall/perimeter recombination losses of single PV cells was dramatically reduced utilizing the ammonium sulfide chemical treatment and subsequent silicon nitride deposition. Monolithic PV arrays were constructed on semi-insulating GaAs substrates. While the semi-insulating GaAs substrate provides a high-resistivity material to facilitate series connection of PV cells, there is still a path for shunt leakage current that can degrade the fill factor, as illustrated in Fig. 5.2 (a). We examined three approaches to investigate

shunt current leakage:

- 1) semi-insulating substrate alone,
- 2) p-GaAs junction barrier (500 nm thick, 10¹⁶ cm⁻³ doping)

3) p-Al_{0.3}Ga_{0.7}As junction barrier (400 nm thick, $5x10^{16}$ cm⁻³ doping).



Figure 5.3. Optical microscope images of two different fabricated PV modules.

We simulated electrical characteristics for each approach using Synopsys Sentaurus TCAD [30] to obtain optimized parameters for layer thickness, p-type doping concentration, and Al mole fraction. We fabricated PV modules with 8 single PV cells (255 µm x 595 µm) connected in series with 10 µm trenches for device isolation, as shown in Fig. 5.3. The 8-cell series connection was designed to achieve a voltage output of approximately 5 V for direct battery charging. The PV module design also incorporated a small integrated photodiode for optical communications, and the possibility to incorporate an open location to mount an external sensor (e.g., pressure) for the system. We measured the electrical characteristics of the PV modules under dark and illuminated conditions using Keithley 2400 and 4200 semiconductor characterization tools. Illumination

utilized a calibrated white light LED or 850 near-infrared (NIR) LED. The incident LED light intensity was approximately $1 \mu W/mm^2$ (420 lux) to simulate a reasonable indoor or subcutaneous low-flux condition that is approximately 1,000 times smaller than AM 1.5 [12] full sun conditions. We studied incident light dependence by varying the irradiance in increments of 10 lux for white light LED illumination and 100 nW/mm² for NIR LED illumination.

5.3. Results

5.3.1. Junction barrier isolation

The *J-V* characteristics of fabricated PV modules with p-GaAs and p-Al_{0.3}Ga_{0.7}As junction barrier isolation are shown in Fig. 5.4 for 850 nm NIR LED illumination at 1.02 μ W/mm² and white LED illumination at 586 lux (1.4 μ W/mm²).





Figure 5.4. Measured J-V characteristics of PV modules with GaAs and AlGaAs barrier layers (a) under 850 nm NIR LED illumination at 1.02 μ W/mm² and (b) under white LED illumination at 586 lux. Comparisons are shown to simulated results (dashed) with shunt leakage removed.

We extracted diode parameters from single PV cells to simulate the *J*-*V* characteristics for series-connected cells without shunt leakage, as shown in Fig. 5.4. PV modules with both p-GaAs and p-Al_{0.3}Ga_{0.7}As junction barrier isolation produced an open circuit voltage of approximately 5 V. This voltage is near the intended design, and is sufficient for direct battery charging without voltage up-conversion. The PV module with p-GaAs junction barrier isolation demonstrated a dramatic degradation in fill factor from 0.754 to 0.463 in comparison to simulated *J*-*V* characteristics neglecting the shunt leakage current. As a result, the overall power conversion efficiency under NIR illumination decreases from the expected value of 28.8 % to 15.0 % due to the inability of the GaAs junction barrier to sufficiently block shunt leakage current. We observed a dramatic improvement in performance by incorporating p-Al_{0.3}Ga_{0.7}As junction barrier isolation,

where the measured power conversion efficiency of 26.3 % under NIR illumination compares favorably to the simulated result of 28.8 % for no shunt leakage.



Figure 5.5. Measured *J-V* curves for varying number of PV cell series connection and (inset) corresponding power conversion efficiency at the maximum power point.

The influence of shunt leakage on the performance of monolithic PV modules is further illustrated by the dependence of J-V on the number of series-connected cells (p-GaAs junction barrier isolation), as shown in Fig. 5.5. The overall power conversion efficiency for NIR illumination demonstrates a clear decrease from 21.4 % to 17.5 %, with a fill factor decreasing from 0.71 to 0.582, as the number of series connections increases from 1 to 7. The degradation in the J-V characteristics shows an obvious shunt leakage characteristic, as represented by the equivalent circuit model shown in Fig. 5.2 (b).

5.3.2. Characteristic of shunt leakage

We measured shunt leakage current between n-contacts of adjacent PV cells without metal interconnects under dark condition for the three device isolation schemes, as shown in Fig. 5.6.



Figure 5.6. Schematic diagram of shunt leakage measurement between bottom n-contacts of adjacent PV cells.





Figure 5.7. Measured shunt leakage current under (a) dark and (b) NIR illumination conditions for three different barrier structures: no barrier, p-GaAs junction, and p-Al_{0.3}Ga_{0.7}As junction.

As shown in Fig. 5.7 (a), we observe an approximately linear *I-V* characteristic for the shunt current leakage without junction barrier isolation, suggesting that it is limited by the resistance of the semi-insulating GaAs substrate. We observe a clear reduction in shunt leakage current by incorporating p-GaAs and p-AlGaAs junction barrier isolation. The shunt leakage *I-V* characteristic for the p-GaAs junction barrier has a near exponential dependence, suggesting that leakage is mediated by the energy barrier height of the p-GaAs junction barrier isolation, which is near the instrument limitation of 1 pA and may be attributed to the increased energy barrier height. The energy barrier height for the three device isolation designs is illustrated in the simulated energy band diagrams shown in Fig. 5.8. Despite the large reduction in Shunt current for the addition of p-GaAs barrier isolation, there is still an obvious degradation in PV module efficiency due to shunt

leakage current (Fig. 5.4). Furthermore, for all three cases of junction barrier isolation, we observe a very low shunt leakage current that is near or below the nA range and would not be expected to dramatically impact PV module efficiency. Therefore, the shunt leakage current characteristics under dark conditions cannot fully explain our observed behavior of the PV modules under illumination.



Figure 5.8. Simulated energy band diagrams between the PV cell base and semi-insulating substrate under dark conditions for three different barrier layer structures: no barrier, p- GaAs junction, and p-Al_{0.3}Ga_{0.7}As junction.

We subsequently measured the shunt leakage current between n-contacts of adjacent single PV cells under illumination and observed a substantial increase in leakage current (Fig. 5.7 (b)). The photo-activated behavior may be interpreted as an undesired increase in photoconductivity at the junction barrier and/or exposed regions of the semi-insulating GaAs substrate. We believe that the nonlinear bias dependence of the photo-activated leakage current may explain the nonlinear shunt leakage current observed in the modules shown in Fig. 5.4. For all three junction isolation leakage techniques, the shunt current increased by a factor of approximately 100. The shunt

leakage for the p-GaAs junction barrier isolation rises to the level of 10 nA, comparable to the photogenerated current, resulting in the observed reduction in PV module performance due to shunt leakage. In contrast, the larger barrier height associated with our p-AlGaAs junction isolation approach significantly improves the ability to block shunt leakage current under illumination. The p-AlGaAs junction barrier isolation limits the shunt leakage current under illumination to approximately 1 nA, preserving the fill factor and overall power conversion efficiency of PV modules.

5.3.3. Comparison of PV module performance to single PV cell

To gauge the overall power generation of the PV modules, we examined the resulting *P-V* characteristics and compared to a 6.4-mm² single PV cell, as shown in Fig. 5.9. The power conversion efficiency of a 1.27-mm² PV module with p-AlGaAs junction barrier isolation was 26.3 % under 850 nm infrared LED illumination at 1.02 μ W/mm² and 16.3 % under white LED indoor conditions at 586 lux (1.4 μ W/mm²). We observed a dramatic decrease in power generation for the p-GaAs junction barrier isolation module, as expected based on the shunt leakage degradation observed in *J-V* characteristics. The power conversion efficiency of the PV module with p-AlGaAs junction barrier isolation, however, approaches the simulated result neglecting shunt leakage current. In comparison to the single PV cell, the PV module with AlGaAs junction barrier isolation provides a voltage that is approximately eight times higher, with approximately 1/8 current reduction.



Figure 5.9. Comparison of measured *P*-*V* characteristics between PV arrays and single PV cell (a) under 850 nm NIR illumination at $1.02 \,\mu$ W/mm² and (b) under white LED illumination at 586 lux.



Figure 5.10. Light intensity dependence of (a) power conversion efficiency and (b) fill factor for single PV and PV modules with p-GaAs and p-AlGaAs barrier junction isolation under NIR illumination.

The voltage generated by the PV module is sufficient for direct battery charging, and provides an effective voltage up-conversion efficiency of approximately 90 % for both NIR and white LED illumination. We observe a strong dependence of power conversion efficiency on

incident light intensity, as shown in Fig. 5.10 for NIR illumination and Fig. 5.11 for White LED illumination.



Figure 5.11. Light intensity dependence of (a) power conversion efficiency and (b) fill factor for single PV and PV modules with p-GaAs and p-AlGaAs barrier junction isolation under white LED illumination.

This dependence has been previously studied for small-area GaAs cells, and attributed to perimeter sidewall recombination. Consistent with our *J*-*V* and *P*-*V* curves, the shunt leakage current for PV modules with p-GaAs barrier junction isolation results in power conversion efficiency degradation over the full range of light intensity studied. The p-AlGaAs barrier junction isolation module demonstrates power conversion efficiency that approaches the performance of the single PV cell for the full range of illumination intensity. These results confirm that the PV module with p-AlGaAs junction barrier isolation can maintain efficiency comparable to a single PV cell under extremely dim light conditions below 100 nW/mm².

5.3.4. Practical application of mm-scale PV energy harvesting

To illustrate the utility of mm-scale PV modules, we constructed a fully-encapsulated 17 mm^3 wireless sensor system incorporating a GaAs PV module (Fig. 5.12) [98]. The system includes a 16 µAh thin-film lithium-ion battery pair that is directly charged by the PV module through a reverse-current blocking diode.



Figure 5.12. Optical microscope images of a wireless mm-scale sensor system with integrated PV module (a) before and (b) after encapsulation.



Figure 5.13. Monitored battery voltage output of this mm-scale system during the charging process under 110 lux indoor illumination.

Battery charging characteristics under dim indoor illumination at 110 lux are shown in Fig. 5.13. The PV module demonstrates a clear recovery of battery voltage within 2 hours, providing adequate energy storage to operate the system. illumination at 110 lux are shown in Fig. 5.12. The PV module demonstrates a clear recovery of battery voltage within 2 hours, providing adequate energy storage to operate the system. The average power requirement to report temperature every 30 minutes is 28.4 nW, comparing favorably to the average power generation of the PV module of 70.8 nW under 200 lux. While the PV system demonstrates the ability to power a mm-scale wireless sensor nodes, there are still opportunities to further improve conversion efficiency through reducing losses that appear at low light intensity (e.g., perimeter recombination) and losses associated with shunt leakage current for modules [18] providing voltage up-conversion.

5.4. Conclusion

Photovoltaic modules offer an efficient means for energy harvesting and direct battery charging in mm-scale systems. This application places unique demands on both PV cells and the module, where PV cells require high performance under much dimmer conditions than conventional solar cells and PV modules have a critical emphasis on maximizing area and electrical isolation of adjacent cells. We demonstrated GaAs PV modules at the mm-scale with high efficiency under low-flux conditions, where AlGaAs junction barrier isolation provided a critical step in limiting shunt leakage current between series connected cells. We observed power conversion efficiency of 26.3 % under 850 nm infrared LED illumination at 1.02 μ W/mm² and 16.3 % under white LED illumination at 586 lux (1.4 μ W/mm²), with a 90 % voltage up-conversion efficiency to reach an operating voltage of 5 V for direct battery charging. We applied a monolithic PV module to demonstrate the perpetual operation of a mm-scale wirelessly interconnected temperature logger system. Further improvements in mm-scale PV module efficiency may be gained by continued improvement in reducing perimeter leakage current in small-area cells and reduction in shunt leakage through techniques such as epitaxial layer transfer to fully insulating substrates [99] or vertical multi-junction designs [100].

CHAPTER 6

Vertical dual junction GaAs Photovoltaics

6.1. Introduction

In previous chapters, we demonstrated single PV cells and monolithic PV modules to directly power fully wireless mm-scale or sub mm-scale systems for IoT and bio-implantable applications or to charge batteries of those systems without DC-DC voltage up-conversion [98]. Series connected single junction (SJ) PV modules at the mm-scale have been demonstrated, providing output voltage of greater than 5 V and voltage up-conversion efficiency of more than 90 %. However, there are fill factor losses associated with shunt leakage paths through the shared substrate [18] and efficiency losses when scaling to small systems due to perimeter losses [13,21]. There is a continuing challenge to miniaturize such PV systems down to the sub mm-scale with minimal optical losses from device isolation and metal interconnects and efficient voltage upconversion. Vertically series connected junctions offer an alternative option for efficient voltage up-conversion, which is commonly used for broadband solar illumination and devices with multiple junctions for differing bandgap energies [101-104]. In this work, we demonstrate dualjunction (DJ) photovoltaic cells at sub mm-scale under low-flux monochromatic illumination (which can also be applied to narrowband visible indoor lighting). The devices use junctions with the same GaAs-based materials and bandgap energies, where cells are designed to evenly divide optical absorption in each junction, while doubling the voltage output for the DJ series connection and considering demanding requirements for miniaturized devices under low-flux operating conditions. The DJ approach provides a means of voltage up-conversion while reducing the number of lateral series connections and corresponding shunt leakage paths for monolithic PV modules.

6.2. Dual-junction device design

6.2.1. Device structure

The conventional multi-junction (MJ) solar cells to boost the power conversion efficiency above 40 % as shown in Fig. 6.1 are utilizing monolithically grown [101-103] or mechanically stacked / bonded [104] materials with optimized band gaps for different spectral contents, where the wider band gap material is placed at the top for the absorption of shorter visible and UV wavelengths and longer wavelength light at the NIR region is absorbed by the lower band gap material below. Monolithic tandem cells based on III-V materials (InGaP/GaAs/InGaAs) [101-103] are preferred due to a higher absorption coefficient and tunability of band gaps using almost lattice matched alloys, where individual cells are connected by heavily doped p-n junctions called tunnel junctions for the series connection of individual cells. Flowing currents of each cell need to be matched through the optimization of device structures due to the series connected device configuration. Advantages of MJ design are to harness more portion of the incident solar spectrum and extract photogenerated electrons with a chemical potential close to incident photons and almost no kinetic / thermalization energy losses, which generates higher power output under the same solar irradiation. The previous laser power converters [105] using vertically-stacked multiple p-n GaAs junctions were designed for the NIR wavelength range between 800 nm and 850 nm.

In this work, we designed two monolithic dual-junction (DJ) PV cells in the same GaAs with optimized thicknesses of each cell based on the structure of conventional III-V tandem cells,

connected by tunnel junction and designed for a 50 % current match for narrow spectral ranges from visible and NIR wavelengths. The AlGaAs window and back surface layers were incorporated to both top and bottom cells as shown in Fig. 6.2.



Figure 6.1. Schematic illustration of (a) monolithically grown and (b) mechanically stacked multi-junction device structures (adopted from [104,106])



Figure 6.2. Schematic diagrams of (a) single dual-junction PV cell and (b) dual-junction PV module illustrating PV cell junction, junction barrier isolation, and shunt leakage path.

6.2.2. Device simulation

Detailed device parameters were optimized utilizing a Sentaurus device simulation tool [30] using well-established physical models for band-to-band tunneling, drift-diffusion currents and photocurrent generation from a transfer-matrix method (TMM). As shown in Fig. 6.3 and Fig 6.4, we initially simulated variable thickness of a top PV cell with the fixed thickness of bottom PV cell at 3.5 μ m under low-flux 1 μ W/mm² NIR illumination at a wavelength of 850 nm, where one-sun solar irradiation is around at 1 mW/mm² [12] and typical indoor lighting is around 400 lux (~ 1 μ W/mm²) [13,21].



Figure 6.3. Simulated current voltage characteristics of dual junction cells with variable thickness of a top cell under 850 nm NIR illumination at $1 \,\mu W/mm^2$.

According to current-voltage (*J-V*) simulation results in Fig. 6.3, there is an evident drop in short circuit current (J_{SC}) of the thinner top cell (100 nm p-type emitter and 150 nm n-type) compared to J_{SC} of the thicker top cell (100 nm p-type emitter and 500 nm n-type base) due to the mismatched current between top and bottom PV cells, degrading the power conversion efficiency from 35.2 % to 18.4 %. Furthermore, we can control the peak wavelength of external quantum efficiency (EQE) above 40 % by controlling thickness of the top cell from orange wavelength at 600 nm targeted for indoor lighting conditions (50 nm p-type emitter and 50 nm n-type base) to NIR wavelength at 850 nm targeted for bio-implantable devices (100 nm p-type emitter and 500 nm n-type base), while having low efficiency off the peak response (short wavelength) due to mismatched absorption between top and bottom junctions.



Figure 6.4. Simulated external quantum efficiency (EQE) spectra for different thickness of top PV cells in the dual-junction structure with the fixed thickness of bottom PV cell at $3.5 \,\mu$ m.

Simulated *J-V* comparison results between dual-junction (DJ) and single-junction (SJ) PV cells under NIR illumination as shown in Fig. 6.5 show that the DJ PV cell has comparable power conversion efficiency (> 30 %) to SJ PV cell with double output voltage (> 1.4 V), which is promising to guarantee the batteryless operation of CMOS chips. Monolithic PV modules utilizing the optimized individual single DJ PV design for higher voltage generation between 5 - 10 V were also considered by incorporating the additional Al_{0.3}Ga_{0.7}As junction barrier (400 nm thick and p-
type doping at 5 x 10^{16} cm⁻³) on the semi-insulating substrate. Optimized device parameters especially for NIR 850 nm wavelength as an example are summarized in Table 6.1.



Figure 6.5. Comparison of simulated *J-V* characteristics between single-junction and dualjunction PV cells under 850 nm NIR illumination at $1 \,\mu$ W/mm².

Туре	Material	Thickness (nm)	Doping (cm ⁻³)	Layer information	
P++	GaAs	200 nm	2x10 ¹⁹	Top PV - anode contact	
P +	Al _{0.8} Ga _{0.2} As	30 nm	2x10 ¹⁸	Top PV - window layer	
P +	GaAs	100 nm	4x10 ¹⁸	Top PV - emitter	
Ν	GaAs	500 nm	1x10 ¹⁷	Top PV – base	
N+	Al _{0.3} Ga _{0.7} As	30 nm	1x10 ¹⁸	Top PV - back surface field	
N++	GaAs	15 nm	1x10 ¹⁹	Tunnel junction	
P++	GaAs	15 nm	4x10 ¹⁹	Tunnel junction	
P +	Al _{0.8} Ga _{0.2} As	30 nm	2x10 ¹⁸	Bottom PV - window layer	
P +	GaAs	500 nm	4x10 ¹⁸	Bottom PV – emitter	
Ν	GaAs	3000 nm	1x10 ¹⁷	Bottom PV – base	
N+	Al _{0.3} Ga _{0.7} As	150 nm	1x10 ¹⁸	Bottom PV - back surface field	
N++	GaAs	1000 nm	2x10 ¹⁸	Bottom PV – cathode contact	
P-	Al _{0.3} Ga _{0.7} As	400 nm	5x10 ¹⁶	Barrier layer	
	GaAs	625 μm	Semi-insulating	Substrate	

Table 6.1. Optimized device parameters of dual junction PV cell and module, designed for NIR

 wavelength at 850 nm.

6.2.3. Device fabrication and testing methods

Wafers based on optimized layer structures (Table 6.1) from the simulation were grown by molecular beam epitaxy on semi-insulating GaAs substates. We fabricated square DJ PV cells ranging from 100 µm to 2.5 mm on a side. Devices were fabricated using conventional photolithography, etching, contact metallization, and liftoff processes, as shown in Fig 6.6. Special attention was devoted to surface/perimeter passivation [13,21] due to their importance for miniaturized PV cell performance, where we used a diluted ammonium sulfide solution treatment [49-51] and a subsequent plasma-enhanced chemical vapor deposition (PECVD) silicon nitride layer [53] that also serves as a top surface anti-reflection coating. Monolithic PV cell modules were fabricated with four and eight connected cells in series using lithographically defined metal interconnects, as shown in Fig. 6.9. These monolithic DJ modules provide voltage up-conversion that combines both vertical and lateral series connections. The detailed device fabrication recipe is included in Appendix.



Figure 6.6. Optical microscope image of fabricated dual junction PV cell at µm-scale.

Electrical characteristics (J-V and P-V) under NIR illumination conditions were measured using a Keithley 4200/2400 parameter analyzer. NIR illumination up to 10 μ W/mm² used a

commercial 850 nm NIR light emitting diode and calibrated power meter. The EQE spectrum was measured using a grating monochromator, xenon white light source, lock-in amplifier, and calibrated photodetector. The measured *J*-*V* and power density versus voltage (*P*-*V*) characteristics were used to extract JSC, open circuit voltage (V_{OC}), maximum power density (P_{max}), and fill factor $FF = P_{max} / (V_{OC}*J_{SC})$.

6.3. Results

6.3.1. Single dual-junction PV cells

Fabricated single DJ PV cells at mm-scale and sub mm-scale were measured under lowflux NIR illumination below 10 μ W/mm² and their device performance parameters especially for fill factor (FF) were extracted Measured single DJ PV cell with 6.4 mm² size achieved power conversion efficiency above 26 % with more than 1.4 V output voltage under 850 nm NIR LED illumination at 2.02 μ W/mm² as shown in Fig. 6.7 while maintaining the extracted fill factor around 0.806 under the dim lighting condition without the degradation from shunt resistance of each junction.



Figure 6.7. Measured *J*-*V* and *P*-*V* characteristics of fabricated dual junction PV cells with 6.4 mm^2 size under 2.02 μ W/mm² NIR illumination at wavelength of 850 nm.

Furthermore, the extremely small size PV cell at μ m-scale (150 μ m x 150 μ m) demonstrated power conversion efficiency greater than 22 % with more than 1.2 V output voltage under higher NIR LED illumination at 6.62 μ W/mm² and produced the output power around 33 nW (1.49 μ W/mm²) as shown in Fig. 6.8, which is sufficient to directly power miniaturized CMOS IC chips [2,8,68] and enables fully remote sensor nodes at μ m-scale [98].



Figure 6.8. Measured *J-V* and *P-V* characteristics of fabricated dual junction PV cells with 150 μ m x 150 μ m size under 2.02 μ W/mm² NIR illumination at wavelength of 850 nm.



Figure 6.9. Optical microscope images of two different fabricated DJ PV modules that have 4 and 8 cells connected in series.

6.3.2. Dual-junction PV modules

Power conversion in monolithic PV modules with four and eight series connected DJ cells are shown in Fig. 6.9, demonstrating higher output voltage that is suitable for direct battery charging capabilities without an inevitable loss during DC-DC voltage up-conversion [92,98]. PV modules with four series connected DJ cells (total of eight diode junctions) in Fig. 6.10 produce an output voltage > 5 V, power conversion efficiency of 23.7 %, and overall voltage up-conversion efficiency of approximately 84 % compared to baseline single-junction PV cells with the output voltage below 1 V, originated from the incorporated AlGaAs junction barrier layer between series connected PV cells. The eight series-connected DJ cell module (total of 16 series connected diode junctions) produces an output voltage > 10 V with an efficiency of more than 18 %, where there is a clear drop in fill factor due to the shunt leakage degradation [98].



Figure 6.10. Measured J-V and P-V characteristics of PV modules with 4 cells in series.



Figure 6.11. Measured J-V and P-V characteristics of PV modules with 8 cells in series.

6.4. Discussion

Device performances of single DJ and SJ PV cells under the same 850nm NIR illumination at 6.62 μ W/mm² were compared as shown in Fig. 6.12 and summarized in Table 6.2. Though there were doubled output voltage (> 1.2 V) and improvement in fill factor above 0.8 as designed in the DJ structure, an undesirable loss of approximately 10 % was observed in photo-generated current (40 % of SJ design, ideally 50 %), results in 6 % efficiency loss. The additional EQE measurement was required to characterize the current loss in the DJ structure more. The EQE measurement of DJ PV cell in Fig. 6.13 indicated that DJ structure was well optimized for the peak wavelength at 850 nm. However, in comparison to measured EQE values of SJ and simulation results of DJ, EQE values at a wavelength of 850 nm were around 38 % for DJ, 85 % for SJ and 45 % for DJ simulation. There was an observed 5 - 7 % EQE loss at 850 nm wavelength, matched to measured lower short-circuit current level of DJ PV cells and requiring more optimization of tunnel junction and the structure of top PV cell.



Figure 6.12. Comparison of measured J-V characteristics between DJ and SJ PV cells under 850 nm NIR illumination at 6.62 μ W/mm².

Parameters	SJ	DJ
$I_{SC} (\mu A/mm^2)$	3.42	1.36
$V_{OC}(V)$	0.71	1.36
$P_{MAX} (\mu W/mm^2)$	1.86	1.49
P _{MAX} (nW)	41.9	33.5
Fill factor	0.766	0.803
η (%)	28.1	22.4

Table 6.2. Comparison of measured device performances between DJ and SJ PV cells under 850 nm NIR illumination at 6.62 μ W/mm².



Figure 6.13. Measured EQE spectra of DJ and SJ PV cells and comparison with simulated EQE spectrum of DJ PV cell.

Furthermore, device performances of SJ and DJ PV modules were compared under 850 nm NIR illumination as shown in Fig. 6.14 and Fig. 6.15. PV modules targeted for above 5 V voltage generation in Fig. 6.14 used total 4 series-connected cells for DJ and 8 series-connected cells for SJ, where the DJ PV module had the improved fill factor around 0.775 than that of the SJ PV module around 0.712 by reducing the number of series connections. However, the power conversion efficiency of the DJ PV module was degraded from 26.8 % to 23.7 %, originated from the photogenerated current loss of individual single DJ PV cells in the module. When both SJ and DJ PV modules used the same 8 cells connected in series as shown in Fig. 6.15, the DJ PV module produced doubled output voltage above 10 V as designed. On the other hand, the measured fill factor of this module was degraded more from 0.712 to 0.618 and corresponding dropped efficiency from 26.8 % to 18.6 % due to higher applied voltage, where the current Al_{0.3}Ga_{0.7}As

barrier layer might be insufficient to block the shunt leakage current through the semi-insulating substrate properly.



Figure 6.14. Comparison of measured *J*-*V* characteristics between SJ (8 cells in series) and DJ (4 cells in series) PV arrays under 850 nm NIR illumination at $2.02 \,\mu W/mm^2$.



Figure 6.15. Comparison of measured *J*-*V* characteristics between SJ (8 cells in series) and DJ (8 cells in series) PV arrays under 850 nm NIR illumination at $2.02 \,\mu$ W/mm².

Parameters	SJ	DJ	
# of cells	8	4	8
I _{SC} (nA/mm ²)	133	112	56.68
V _{OC} (V)	5.61	5.46	10.62
$P_{MAX} (nW/mm^2)$	541	479	376
Fill factor	0.712	0.775	0.618
η (%)	26.8	23.7	18.6

Table 6.3. Comparison of measured device performances between SJ and DJ PV modules with 4 and 8 cells in series under 850 nm NIR illumination at 6.62 μ W/mm².

Further improvements in the performance of DJ PV cells and modules are possible from further optimization of the tunnel junction and layer structure of the top junction to reduce photogenerated current losses. The shunt leakage current of PV modules through the substrate can be reduced using thicker junction barrier layers with wider bandgap AlGaAs (> 30 % aluminum) or techniques such as epitaxial lift-off (ELO) [103,107] to transfer the active layers to insulating substates or vertically-stacked multi-junction designs [105] rather than laterally interconnected cells.

6.4. Conclusion

Vertically stacked DJ PV cells and modules are demonstrated to increase operating voltage for direct powering of miniature devices for IoT and bio-implantable applications with lowirradiance narrowband spectral illumination. The DJ approach increases the output voltage per cell and minimizes area losses from device isolation and interconnects in comparison to SJ cells. DJ PV cells at small dimensions (150 μ m x 150 μ m) demonstrate power conversion efficiency greater than 22 % with more than 1.2 V output voltage under low-flux 850 nm NIR LED illumination at $6.62 \,\mu$ W/mm², which is sufficient for batteryless operation of miniaturized CMOS IC chips. The output voltage of DJ PV modules with four series-connected single cells was greater than 5 V while maintaining an efficiency of more than 23 %. Further power conversion efficiency improvements are expected by optimizing designs to minimize photocurrent collection losses and shunt resistance losses through the substrate in modules. In addition to monochromatic NIR illumination, the GaAs DJ approach also shows promise for efficient energy harvesting under narrowband artificial indoor lighting conditions.

CHAPTER 7

Monolithically Integrated Microscale LED and Photovoltaic Module

7.1. Introduction

Neural recording, and ultimately the achievement of brain machine interfaces [84,108-112], ultimately relies on truly floating, wireless neural recording implants that do minimal damage and enable high channel count (> 1,000). Wireless power delivery and data communications are critical, though the small dimensions required (10's to low 100's of micrometers) and constraints placed by surrounding biological tissue make conventional wireless approaches highly inefficient. Radiofrequency (RF) links suffer from dramatic antenna efficiency loss when scaled to sub-mm size [74,75] due to lateral and angular misalignments [76] and weak coupling [75] with mm-scale antenna receivers or require high frequency (> 50 GHz), which results in tissue absorption and high power consumption [113-115]. Ultra-sound offers a path to high efficiency wireless power transfer comparable to RF but faces difficulty scaling to sub-mm size due to bulky transducers [116]. Near-Infrared light (NIR, 650 - 1350 nm) offers a transparency window for biological tissue [80,117] and device technologies with high efficiency at the micrometer scale. A system on a chip based for recording neural activities has been demonstrated using NIR for wireless power and data communications [118,119]. This device used an AlGaAs diode that operated simultaneously as a photovoltaic (PV) cell for power generation and light emitting diode (LED) for data communications. However, the input irradiance around 100 mW/mm² to operate the chips was far

much higher than the maximum exposure limit at 1.36 mW/mm^2 from ANSI [82], which might cause the damage to the brain due to the temperature increase under the intensive illumination at visible wavelength region. Also, the energy harvesting of PV cell was interrupted during the light pulse generation for the communication. Scaled microscale LEDs (μ -LEDs) on the order of 50 μ m square have demonstrated the ability for scaled implantable devices for optogenetics [120], though operating at visible wavelengths and at lower power density that would not meet demands for neural dust.

We propose an approach for wireless neural recording based on arrays of implanted microscale probes (motes) with a NIR link to a repeater unit that would be inductively linked to an external device as shown in Fig. 7.1. Each mote would consist of a non-invasive cerebral implant and an integrated and packaged CMOS / optoelectronic device floating on the pia. Each mote would communicate through the arachnoid and dura mater to a cranial implant repeater unit that would be inductively coupled to an external device.



Figure 7.1. Conceptual illustration of a full system for recording and transmitting the neural signals and a PV / LED module in the system for NIR wireless power transfer and data communications (Drawn by Jongyup Lim).

In this work, we specifically focus on the "last millimeter" problem for meeting the wireless power and data communications requirements between the motes and the repeater unit, utilizing monolithically integrated GaAs / InGaAs based PV and μ -LED module at μ m-scale. We used 850 nm NIR light for wireless power transfer and downlink data communication to the motes, and 1000 nm NIR light for uplink communication from the motes to the repeater. We selected the shorter wavelength ($\lambda = 850$ nm) for power generation to maximize PV power and voltage generation to the CMOS circuitry. In turn, the longer wavelength for the data communication uplink ($\lambda = 1000$ nm) reduces the minimum required voltage to drive the μ -LEDs. As a result, the μ -LED driver circuits can operate directly on the PV cell voltage without the need for voltage up conversion, thereby reducing the circuit complexity, power loss and CMOS layer size. Separate wavelengths also provide a means to distinguish the two optical signals.

7.2. Optical transmittance of dura maters

The optical transmittance of NIR light through the dura mater [80,117] is a key point for the feasibility of the proposed device concept for the NIR optical link, determined by optical absorption and scattering in the tissue due to the structural elements (collagen fibrils) and base substance (interstitial fluid). We measured NIR transmission through the dura of the post-craniotomy feline and a non-human primate (NHP) using the NIR LED and laser sources aligned to a calibrated photodetector as shown in Fig. 7.2. The dura showed high optical transmittance of > 85 % for the thinner feline dura under 850 nm NIR laser illumination and > 35 % for the thicker NHP dura under 850 nm, 970 nm and 1050 nm NIR LED illumination as show in Fig. 7.3 and Fig. 7.4, which is well matched to previously reported results [80,117]. As a result, we estimated the transmittance for humans slightly more conservative at 30 %, which is sufficient to guarantee

the operation of NIR based PV and LED modules.



Figure 7.2. Experiment set up for measuring the NIR optical transmittance through the dura samples using laser and LED illumination sources.



Figure 7.3. Measured NIR transmittance through post-craniotomy feline dura samples under 850 nm laser illumination.



Figure 7.4. Measured NIR transmittance through post-craniotomy NHP dura samples under 850 nm, 970 nm and 1050 nm NIR LED illumination.

7.4. PV / LED module structure and fabrication

We designed monolithically integrated GaAs / InGaAs based on PV and LED nodule at μ m-scale as shown in Fig. 7.5, utilizing 850 nm NIR light for wireless power transfer to PV under low-flux condition and 1000 nm NIR light emission form μ LED for uplink communication. We fabricated the module with μ m-scale size (190 μ m x 170 μ m for PV and 50 μ m x 50 μ m for LED) as shown in Fig. 7.6 using the conventional fabrication processes, where the chlorine based reactive-ion etching (RIE) to etch the epitaxial AlGaAs / GaAs / InGaAs layers down to desired contact layers both for PV and LED by controlling the etch depth using the surface profiler, the conventional photolithography using photoresists, plasma enhanced chemical vapor deposition (PECVD) of insulating films for passivation of the exposed sidewall, and physical vapor deposition (PVD) of thin metal films were used. The PV cell incorporated a vertically-stacked, dual-junction configuration optimized for a wavelength of 850 nm (The detailed device structure

was discussed in Chapter 6). The series connection of the two junctions doubles the voltage output, which can achieve sufficient voltage for direct powering of a CMOS chip with the minimum voltage requirement near 1.4 V [115] while also avoiding critical area and power losses due to metal interconnects and DC-DC voltage up-conversion. Two monolithic PV cells connected by tunnel junction were designed for a 50 % current match at a NIR wavelength of 850 nm by controlling the thickness of each cell. The In_{0.2}Ga_{0.8}As multi quantum-well (QW) μ -LED for NIR data links monolithically integrated with the PV cell, designed for emission at 1000 nm.



Figure 7.5. Schematic diagram of monolithically integrated μ -LED / PV module illustrating tandem junction of PVs and junction barrier isolation between LED and PV.



Figure 7.6. Optical microscope image of fabricated LED / PV module at µm-scale.

7.5. PV / LED module performance

7.5.1. Photovoltaic energy harvesting

A fabricated PV cell (190 μ m x 170 μ m) as shown in Fig. 7.7 demonstrated power conversion efficiency greater than 25 % and provides ~ 1 μ W with more than 1.4 V output voltage under 850 nm NIR LED illumination at 100 μ W/mm² (6 x lower than tissue exposure limit of 640 μ W/mm²), which is sufficient to directly power miniaturized CMOS IC chips subcutaneously without the concern of tissue heating above 2 °C [84].



Figure 7.7. Measured *I-V* and *P-V* characteristics of dual junction PV cell under 850 nm NIR LED illumination at $100 \,\mu$ W/mm².

7.5.2. Micro-LED data-communications link

The NIR data link between the motes and the Repeater is designed to use a μ -LED on the motes and a single-photon avalanche detector (SPAD) array [121] at the Repeater, operating at a wavelength of 1000 nm. This wavelength provides high transmittance through the dura, high

quantum efficiency for InGaAs based photodetectors, and minimizes voltage and power requirements to fire the μ -LED. The use of a SPAD [122] – a photodetector with internal gain operating in Geiger mode – provides a highly sensitive means of detecting signals from the motes, where only a few photons at the receiver are needed to trigger a response. The photons emitted from the LED should exceed the dark count rate of a SPAD on the Repeater, given by the following equation (1)

$$N_{LED} > \frac{(DCR)(F)(\Delta t)}{\eta_{dura}\eta_{SPAD}}$$
(1)

where η_{dura} is the number of photons received at the SPAD relative to number of photons emitted by the LED, η_{SPAD} is the detection efficiency of the SPAD, *DCR* is the SPAD dark count rate, *F* is the noise margin (multiplier) for number of photons above *DCR* desired to achieve the desired signal-to-noise ratio, and Δt is the time window. Monte Carlo simulation [123] of NIR emission from the microscale LED through dura is shown in Fig. 7.8, accounting for tissue absorption and scattering [80,117], and Lambertian distribution of light output from a planar LED device.



Figure 7.8. Monte-Carlo simulation of NIR emission from a μ -LED through dura to the SPAD repeater, with estimated efficiency of 0.1% (photons received at SPAD with 100 μ m diameter relative to photons emitted from LED) (Simulated by Michael Barrow)

The efficiency of this optical link is estimated to be $\eta_{dura} = 0.1$ %. Assuming conservative values typical of commercial SPADs, $\eta_{SPAD} = 10$ %, DCR = 5,000 counts / second; a noise margin of F = 100; the LED is required to provide $N_{LED} > 1x10^6$ photons for a time window corresponding to $\Delta t = 1/DCR = 200 \ \mu$ s. This corresponds to an LED optical power of 1 nW in a 200 μ s pulse, where alternatively shorter pulse widths and higher LED power can be scaled to ensure detection at the Repeater. The fabricated μ -LED (50 μ m x 50 μ m) showed exponential *I-V* characteristics with the turn-on voltage around 1 V (below the output voltage of PV cell) as shown in Fig. 7.9 and a clear emission peak at a wavelength of 1000 nm from the photoluminescence (PL) result in Fig. 7.10. The emitted light output from LED along with the applied DC current was measured by the calibrated photodetector with 1 cm diameter, showing that μ -LED meets required optical power (> 1 nW) for wireless two-way communications when the applied current is above 10 μ A as shown in Fig. 7.11.



Figure 7.9. Measured *I-V* characteristic of fabricated LED with 50 µm x 50 µm size.



Figure 7.10. Measured photoluminescence (PL) versus wavelength curve of fabricated LED with $50 \,\mu\text{m} \text{ size}$.



Figure 7.11. Measured light output versus applied current curve of fabricated LED with 50 μ m x 50 μ m size.

We also demonstrated the practical pulse detection with the SPAD. The applied input pulses to LED with 40 μ s pulse width and 1.54 V amplitude (corresponding LED current ~ 839 μ A) was clearly detected by the SPAD with 10 μ s dead time as shown in Fig. 7.12, which is promising for the NIR based data up-link.



Figure 7.12. Detected SPAD output (10 µs deadtime) versus LED input pulses (40 µs pulse width, 100 µs period, 1.54 applied voltage) (Collaborating with Jongyup Lim).

7.6. Discussion

7.6.1. PV and LED isolation

We successfully demonstrated that the fabricated GaAs PV cell (190 μ m x 170 μ m) using the dual-junction design generated ~ 1 μ W with more than 25 % power conversion efficiency and above 1.4 V output voltage under 850 nm NIR LED illumination at 100 μ W/mm² (6 x lower than tissue exposure limit), which is sufficient to guarantee the batteryless operation of miniaturized CMOS IC chips under subcutaneous conditions. For the operation of a stacked full system, the PV / μ -LED module needs to be connected to the CMOS control circuity using the through-wafer vias and interconnects [124,125] or metal wires from the wire-bonding process. The connection of the PV / μ -LED module to CMOS circuitry requires shared cathodes between devices, requiring electrical isolation to avoid degradation due to shunt leakage current through the device stack at the required operating voltages. After sharing cathodes, the output voltage of PV cell was degraded around 1.0 V as shown in Fig. 7.13 due to poor isolation, which is problematic to operate the full system without the additional charge storage layer. The current p-type Al_{0.3}Ga_{0.7}As barrier with 100 nm thickness and 5×10^{16} cm⁻³ doping was not sufficient to block the leakage current, where the leakage current between LED cathode and PV anode showed the clear diode behavior as shown in Fig.7.14. In this manner, we defined an isolation trench with 10 µm wide and 7 µm deep between LED and PV by etching down to the semi-insulating GaAs substrate with high resistivity around 10 MΩ-cm. The isolation trench between PV and µLED blocked the leakage current successfully below 10's nA range (Fig. 7.14) even under NIR LED illumination, which makes the output voltage of PV cell recovered to its originally designed value (> 1.4 V) as shown in Fig. 7.15. Though this isolation method was working properly without the additional modification of current epi-structure, this method made unwanted area losses and increased the fabrication complexity for patterning the PR and depositing the interconnects due to deep trench with the poor aspect ratio. A new epi-structure using the AlGaAs based p-n junction isolation layers between PV and µ-LED might be helpful to block the leakage current by increasing the barrier height.



Figure 7.13. Measured *I-V* characteristics of PV cell before / after the shared connection to LED cathode under 850 nm NIR LED illumination at 150 μ W/mm².



Figure 7.14. Measured leakage current between LED cathode and PV anode with / without the isolation trench.



Figure 7.15. Measured *I-V* characteristics of PV cell with / without the isolation trench under 850 nm NIR LED illumination at $150 \,\mu$ W/mm².

7.6.2. µ-LED performance

The monolithically integrated In_{0.2}Ga_{0.8}As QW μ -LED (50 μ m x 50 μ m) showed good exponential *I-V* characteristics and emitted the enough light (> 1 nW) with the center wavelength

at 1000 nm for NIR data link. The fabricated LED demonstrated the example detection of digital pulses (pulse width ~ 40 μ s, period ~ 100 μ s, applied LED current amplitude ~ 800 μ A) using the highly sensitive SPAD for wireless two-way communications. However, the system operation using shorter pulses below 100 ns is preferred for the energy efficient detection at the SPAD, where the current LED might be problematic because the LED requires more than 1 mA applied current to emit the optical power > 1 μ W as shown in Fig. 7.11. This low-intensity light output can be explained by the extracted external quantum efficiency (EQE) from the *L-I* curves. The extracted EQE was below 0.1 %, which is far much lower than simulated EQE results ~ 1 % assuming the 50 % internal quantum efficiency (IQE) and 2 % extraction efficiency through the planar surface. The EQE measurements for variable area μ -LEDs were done to characterize the sources of poor EQE of μ -LED as shown in Fig. 7.16. Variable area μ -LEDs showed clear degradation in EQE below 100 μ m x 100 μ m due to the possible increase in sidewall / perimeter recombination losses, mitigated by additional sidewall passivation [49-51] and changing pad geometry for current spreading.



Figure 7.16. Extracted EQE versus applied current curves of variable size LEDs.

The temperature dependent PL measurement was done for the InGaAs epi-wafer to estimate the IQE, showing that the clear blue shift of peak wavelengths and the gradual improvement in the PL intensity were observed as the temperature was decreased from 300 K to 13 K as shown in Fig. 7.17. However, the rough extraction of IQE by comparing the peak intensity at 13 K to the peak intensity at 300 K was around 10 %, which means that InGaAs QW material is relatively fine and these is not a major issue in epitaxial structure and quality of growth.



Figure 7.17. (a) Measured temperature dependent photoluminescence spectra of InGaAs LED and (b) extracted peak intensity values versus variable temperature from 300 K to 13 K (collaborating with Anthony Aiello).

In summary, main reasons for the low EQE of InGaAs LED are connected to poor light extraction efficiency through the planar surface, current spreading and dead area near sidewall. Further improvements in LED performances are possible using the surface texturing or roughing [126] to improve the extraction efficiency, modifying the cladding layer design for better light extraction and current spreading, and mitigating the sidewall / perimeter recombination losses from the chemical and dielectric passivation studies.

7.7. Conclusion

Our monolithic PV / μ LED module demonstrates the viability of near-infrared light to meet power and data communications requirements for devices at dimensions desired for bioimplantable systems such as "neural dust". The dual junction GaAs PV cell provides sufficient power (~ 1 μ W) and voltage (> 1.4 V) for battery-less operation of CMOS circuitry under 850 nm NIR illumination at 100 μ W/mm² within the tissue exposure limit. The monolithically integrated InGaAs μ -LED emits NIR light pulses at detectable power levels (> 1 nW) under realistic energy harvesting levels from PV cells, where we have demonstrated example digital pulse detection with a single photon avalanche photodiode (SPAD).

CHAPTER 8

Future work and conclusion

8.1. Future work

8.1.1. 3D vertical integration

The proposed system (in Chapter 7) for recording and transmitting the neural signals through the NIR optical link requires the wafer-level vertical integration of the GaAs PV / LED module and CMOS silicon chips as shown in Fig. 8.1, while maintaining a total thickness below 150 μ m. The integration processes [124,125] are including the through-wafer via (TWV) processing, wafer thinning, flip chip bonding, laser dicing, neural probe insertion, packaging and die singulation. TWVs are defined both for GaAs (Fig. 8.2) and silicon wafers using the wellestablished deep reactive ion etching (DRIE) recipes (chlorine based etch for GaAs [127] and fluorine based etch for silicon [128]), deeper than the target thickness of wafer thinning in order to expose the via after the wafer thinning process. Next, a 30 nm Al₂O₃ for the insulating layer and subsequent a 10 nm Pt for the conducting layer are deposited via atomic layer deposition (ALD), which provides a conductive path through the TWVs while preventing shorting to the substrate. The patterned Pt layer is used as a seed layer for electrochemical deposition (ECD) of copper to fill the TWV, which improves the conduction of TWVs. A parylene layer with 20 µm thickness is deposited prior to thinning for the mechanical support of thinned GaAs and silicon wafers around 50 µm thickness. After the thinning process, back contacts on the thinned GaAs wafer are defined using the sputtered films with 10 nm / 500 nm Ti / Au. For silicon chips, the additional holes for the insertion and the electrical connection of neural probes are etched on the backside of the chips using DRIE and deposit the insulating Al_2O_3 and the conducting Pt again using ALD. Both flexible GaAs and silicon wafers are then mounted to the rigid silicon carrier wafers and attached together using the flip chip bonding process with 30 µm solder balls on the back contacts of GaAs and the front contacts of silicon followed by the alignment of chips and a solder reflow. Once two chips are bonded together, the assemblies are fully delineated by laser dicing, where the diced assemblies are mounted temporally on the tape for the future packaging process. The neural probes are inserted to the assemblies using the silver paint for the electrical connection. The fully assemblies are passivated by bio-compatible packaging methods for multi-year implantation. \setminus



Figure 8.1. Cross-sectional concept diagram of the integration of GaAs PV / LED modules, silicon chips and neural probe.



Figure 8.2. Schematic diagrams for (a) the front side and (b) back side of GaAs PV / LED module for the vertical integration, illustrating the through wafer via, interconnects and back contacts.

Initial results for TWVs for GaAs (Fig. 8.3 as an example) and silicon showed the welldefined TWVs with 50 μ m deep and 20 μ m wide, where the exposed sidewall was covered by the insulating 30 nm Al₂O₃ and subsequent conducting 10 nm Pt layers.



Figure 8.3. (a) Microscope and (b) scanning electron microscopy (SEM) images of through wafer vias of GaAs, deeper than 50 µm.

The wafer thinning process followed by Ti / Au back contact deposition on the thinned wafers were successfully done as shown in Fig. 8.4, where the TWVs were well exposed and the electrical connection between top and bottom contacts through the thin ALD conducting layer with a resistance of 685 Ω was observed as shown in Fig. 8.5.



Figure 8.4. (a) An image of the thinned GaAs wafer mounted to the flexible tape and (b) a microscope image of the backside of thinned wafer, exposing the TWVs and predefined alignment marks.



Figure 8.5. (a) An optical microscope image of backside of thinned GaAs with TWVs, and (b) electrical characteristics measured through two GaAs TWVs in series (collaborating with Michael Barrow).

Further process development and improvement will be required to make the full assemblies in terms of via backfill using the ECD copper, electrical properties of conducting layers through TWVs, flip chip bonding, carrier wafer handling, laser dicing and packaging.

8.1.2. AlGaAs double heterojunction LED

A repeater unit (RU) in the epidural space (discussed in Chapter 7) for the wireless communication with the integrated assemblies of GaAs PV / LED modules, silicon chips and the external unit (EU). The RU emits modulated 850nm NIR light for power transfer to PV and data downlink to the CMOS chips. The RU receives the neural recording signals through the dura mater using an array of 1000 nm NIR SPADs, where the current single InGaAs / InP based SPADs [121,129] show reasonable detection performances above 10 % quantum efficiency ($\lambda = 900$ nm \sim 1700 nm) that is sufficient to operate the current system. However, InGaAs / InP based SPADs have some issues to make an array structure at mm-scale size for our target of 1000 + channel count, requiring the integrated cooling system and yield issues [121,129]. On the other hand, the CMOS SPAD arrays [122,130], at mm-scale are already developed with the operating wavelength range from 340 nm to 900 nm as shown in Fig. 8.6, which means that our emission wavelength at 1000 nm from InGaAs LED is not adequate for the wireless communication to the CMOS SPAD array. In this manner, we switch the target wavelength to 780 nm because this wavelength has good transmittance of human dura mater (> 30 %) [80,117], selective to the powering wavelength at 850 nm to GaAs PV and the well-established LED structures using AlGaAs based double heterojunction (DH) [131,132] without the concern for the lattice mismatch. The proposed AlGaAs DH device structure is shown in Fig. 8.7. The active Al_{0.12}Ga_{0.88}As layer cladded by larger bandgap Al_{0.3}Ga_{0.7}As layers improves the carrier injection efficiency and reduces non-radiative

recombination losses from the surface. The top p+ AlGaAs layer is thick around 1 μ m for the better current spreading.



Figure 8.6. Photon detection probability along with wavelengths at different excess bias at room temperature (adopted from [122]).



Figure 8.7. Optimized device parameters of DH AlGaAs LED with the emission wavelength at 780 nm.

The simulation results of optimized DH AlGaAs LED compared to the original InGaAs LED as shown in Fig. 8.8 show good current-voltage characteristics with the turn-on voltage above 1.5 V while maintaining high IQE above 80 % at our operating regime, which is promising for our future repeater designs using CMOS SPAD arrays for the operation of 1000 + count neural probe assemblies.



Figure 8.8. (a) Simulated current versus voltage characteristics and (b) IQE versus voltage curves of DH AlGaAs LED, compared to InGaAs QW LED.

8.1.3. Multi-junction PV designs for visible light

The data of dual-junction (DJ) PV cells and modules thus far has focused on a monochromatic wavelength of 850 nm, providing an important spectral region for bio-implantable devices or laser power converters. Extending the DJ approach to indoor lighting applications requires designs that shift the peak EQE towards visible wavelengths. This may be accomplished by either reducing the top junction thickness (Fig. 6.4), or increasing the bandgap energy for the optical absorption regions of the device. Reducing thickness of a top GaAs junction is not an optimal solution since the bandgap energy of GaAs material at 1.424 eV is not perfectly adequate for the indoor lighting spectrum, requiring wider bandgap around 1.9 eV [15,21]. Increasing aluminum content in the top GaAs absorbing region can be an effective approach to shift peak EQE to the visible spectral region while increasing the bandgap energy close to the ideal bandgap energy for the indoor lighting spectrum. Simulated EQE spectra for DJ designs with varying aluminum mole fraction in top junctions are shown in Fig. 8.9, demonstrating a shift to visible wavelengths. High EQE values are reached in the 600-650 nm region, corresponding to high flux regions for the soft white artificial lighting with 3000 K color temperature. Simulated J-V characteristics for DJ designs with GaAs and Al_{0.4}Ga_{0.6}As absorber top junctions are shown in Fig. 8.10 for illumination with a soft white LED. The use of higher aluminum mole fraction in the top cell results in a substantial increase in power conversion efficiency with values exceeding 30 %. The incorporation of a wider bandgap top junction also provides a higher output voltage due to the more optimal match to the ideal bandgap energy for the indoor lighting spectrum [15,21].



Figure 8.9. Simulated EQE spectra for variable aluminum mole fraction of top PV cells in the DJ structure with the fixed thickness of top PV cell at 400 nm and comparison with the normalized spectrum of commercialized white LED with 3000 K color temperature.



Figure. 8.10. Simulated *J-V* characteristics of dual junction cells with variable aluminum mole fraction of a top cell under white LED illumination (3000 K color temperature) at 1μ W/mm².
8.2. Conclusion

This thesis covers an in-depth study of photovoltaic energy harvesting for the Internet of Things (IoT) and bio-implantable devices based on mm-scale sensors, which can be applied to a variety of applications including biomedical devices, surveillance, micro-robots and industrial monitoring. Energy harvesting approaches to power miniaturized mm-scale sensors have traditionally utilized thermal energy, mechanical vibrations and radio frequency electromagnetic radiation. However, the achievement of efficient energy harvesting at the mm-scale or sub mmscale has been elusive.

In this work, I show that photovoltaic cells and modules at the mm-scale or sub mm-scale can be an alternative means of wireless power transfer to mm-scale sensors for IoT and bioimplantable devices, utilizing ambient indoor lighting (425-650 nm) or intentional irradiation of near-infrared LED sources (700-1100 nm) through biological tissue. Dark current and shunt conductive paths in photovoltaic cells become much more important at these small dimensions and low flux illumination conditions in comparison to typical outdoor solar irradiation (approximately a factor of 1,000 lower flux than AM 1.5).

Single silicon and GaAs photovoltaic cells at the mm-scale can achieve a power conversion efficiency of greater than 17 % for silicon and 30 % for GaAs under 850 nm NIR irradiation at $1 \,\mu$ W/mm² through the optimized device structure and sidewall/surface passivation studies, which guarantees perpetual operation of mm-scale sensors. These photovoltaic cells demonstrate highly efficient energy harvesting through various biological tissue samples from ambient sunlight, or irradiation from infrared sources. Sufficient power generation above 50 nW/mm² is achieved for perpetual operation of mm-scale systems for implant depth of at least 15 mm including hair / skin / muscle / bone under 850 nm NIR illumination at 134 μ W/mm² within the maximum tissue exposure limit. Furthermore, monolithic single-junction GaAs photovoltaic modules offer an

efficient means for voltage up-conversion to charge battery directly in mm-scale systems. I demonstrate SJ GaAs PV modules at the mm-scale with high efficiency under low-flux conditions, where AlGaAs junction barrier isolation provided a critical step in limiting shunt leakage current between series connected cells. I observed power conversion efficiency of great than 26 % under 850 nm infrared LED illumination at 1 μ W/mm² and 16 % under white LED illumination at 586 lux (1.4 μ W/mm²), with a 90 % voltage up-conversion efficiency to reach an operating voltage of 5 V for direct battery charging. However, there are fill factor losses associated with shunt leakage paths through the shared substrate and efficiency losses when scaling to small systems due to perimeter losses. There is a continuing challenge to miniaturize such PV systems down to the sub mm-scale with minimal optical losses from device isolation and metal interconnects and efficient voltage up-conversion.

Vertically stacked DJ PV cells and modules are demonstrated to increase operating voltage for direct powering of miniature devices for IoT and bio-implantable applications with lowirradiance narrowband spectral illumination. The DJ approach increases the output voltage per cell and minimizes area losses from device isolation and interconnects in comparison to SJ cells. DJ PV cells at small dimensions (150 μ m x 150 μ m) demonstrate power conversion efficiency greater than 22 % with more than 1.2 V output voltage under low-flux 850 nm NIR LED illumination at 6.62 μ W/mm², which is sufficient for batteryless operation of miniaturized CMOS IC chips. The output voltage of DJ PV modules with eight series-connected single cells was greater than 10 V while maintaining an efficiency of more than 18 %. Further power conversion efficiency improvements are expected by optimizing designs to minimize photocurrent collection losses and shunt resistance losses through the substrate in modules. In addition to monochromatic NIR illumination, the GaAs DJ approach also shows promise for efficient energy harvesting under narrowband artificial indoor lighting conditions.

Finally, monolithic PV/LED modules at the sub mm-scale for brain-machine interfaces are demonstrated, enabling two-way optical power and data transfer in a through-tissue configuration. The DJ GaAs PV cell provides sufficient power (~ 1 μ W) and voltage (> 1.4 V) for batteryless operation of CMOS circuitry under 850 nm NIR illumination at 100 μ W/mm² within the tissue exposure limit. The monolithically integrated InGaAs μ -LEDs emit NIR light pulses (1000 nm) at detectable power levels (> 1 nW) under realistic energy harvesting levels from PV cells, where we have demonstrated example digital pulse detection with a single photon avalanche photodiode (SPAD). The wafer-level assembly plan for the 3D vertical integration of three different systems including GaAs LED/PV modules, CMOS silicon chips, and neural probes is proposed, using the through-wafer vias, wafer thinning, flip chip bonding, laser dicing, and final packaging.

Appendix

Fabrication Recipes for Silicon and GaAs Photovoltaic Cells

1. Silicon Photovoltaic cells

1.1. Surface n-type doing

- 1.1.1. RCA clean
- 1.1.2. Phosphorous doping (S5T1) : POCl₃ at 975 °C

(Deposition : 12 min, Anealing : 10 min), PSG thickness 80nm

1.1.3. Measured sheet resistance: 9.65 Ω / sq (epi-wafer), 8.8 Ω / sq

(Monitor wafer: Sheet resistance without surface doping, 755 Ω / sq)

1.1.4. BHF for PSG etch (After the etching, the surface should be 'Hydrophobic')

1.2. Shallow device isolation #1

- 1.2.1. Surface clean: Nanostrip for 15 min
- 1.2.2. Dehydration bake: 150 °C, 10 min
- 1.2.3. PR coating (ACS200): SPR 3.0, 2000 rpm, 3 um thick,
- 1.2.4. PR exposure (MA6): 8 s exposure time, hard contact, 30um gap
- 1.2.5. Post exposure bake (ACS200): 115 °C, 90 s
- 1.2.6. PR development (ACS200): 60 s, Develop 300DEV (Recipe)
- 1.2.7. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.2.8. Back surface cleaning: Acetone, IPA clean
- 1.2.9. DRIE (STS4): 40 s etch time, 3.4 µm target etch depth

1.2.10. PR strip (Yes plasma stripper): 800 W, 400 s, 150 °C, 80 sccm O₂

1.2.11. Wafer cleaning : 5 min Acetone, 5 min IPA

1.2.12. DI water Spin / Rinse / Dry

1.3. Deep device isolation #2

- 1.3.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.3.2. DI water Spin / Rinse / Dry
- 1.3.3. Dehydration bake: 150 °C, 10 min
- 1.3.4. PR coating (ACS200): SPR 3.0, 2000 rpm, 3 um target,
- 1.3.5. PR exposure (MA6): 8 s exposure time, hard contact, 30 µm gap
- 1.3.6. Post exposure bake (ACS200): 115 °C, 90 s
- 1.3.7. PR development (ACS200): 60 s, Develop 300DEV (Recipe)
- 1.3.8. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.3.9. Back surface cleaning: Acetone, IPA clean
- 1.3.10. DRIE (STS4): 7 min 30 s etch time, 44 µm target etch depth
- 1.3.11. PR strip (Yes plasma stripper): 800 W, 400 s, 150 °C, 80 sccm O₂
- 1.3.12. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.3.13. DI water Spin / Rinse / Dry

1.4. Etch heavily doped surface layer ~ 300 nm

- 1.4.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.4.2. DI water Spin / Rinse / Dry
- 1.4.3. Dehydration bake: 150 °C, 10 min
- 1.4.4. HMDS deposition (Yes image reversal oven)
- 1.4.5. PR coating (CEE spinner): SPR 7.0, 2000 rpm, 10 µm thick

1.4.6. Soft bake (Hot plate): 85 °C, 330 s

1.4.7. PR exposure (MA6): 30 s exposure time, hard contact, 30 µm gap

1.4.8. Post exposure delay: 3 hours to finish the photochemical reaction

1.4.9. Post exposure bake (Hot plate): 85 °C, 90 s

1.4.10. PR development (CEE developer): AZ726, DP 80-80

1.4.11. Plasma ashing (Yes plasma stripper): 100 W, 200 s, 35 sccm O₂

1.4.12. DRIE (P5000): 1 min etch time, 300 nm target etch depth,

Poly PAT BKM (Recipe, 40 sccm HBr, 40 sccm Cl₂)

- 1.4.13. PR strip (Yes plasma stripper): 800 W, 400 s, 150 °C, 80 sccm O₂
- 1.4.14. Wafer cleaning: 5 min Acetone, 5 min IPA

1.4.15. DI water Spin / Rinse / Dry

1.4.16. Residue removal: 5 min BHF

1.5. Passivation layer deposition

- 1.5.1. RCA clean
- 1.5.2. Dry oxide deposition (S3T1): 15 min growth at 900 °C, 40 nm target
- 1.5.3. Amorphous silicon deposition (S3T3): 40 min growth at 560 °C,

25 nm target thickness, 4.35 refractive index

1.5.4. LPCVD Si₃N₄ (S2T3): 10 min growth at 800 °C, 50 nm target thickness

1.6. Metal openings – Top

- 1.6.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.6.2. DI water Spin / Rinse / Dry

1.6.3. Dehydration bake: 150 °C, 10 min

1.6.4. HMDS deposition (Yes image reversal oven): 300 s

- 1.6.5. PR coating (CEE spinner): SPR 7.0, 1000 rpm, 15 µm thick
- 1.6.6. Soft bake (Hot plate): 85 °C, 600 s
- 1.6.7. PR exposure (MA6): 45 s exposure time, hard contact, 30 µm gap
- 1.6.8. Post exposure delay: 4 hours to finish the photochemical reaction
- 1.6.9. Post exposure bake (Hot plate): 85 °C, 600 s
- 1.6.10. PR development (CEE developer): AZ726, DP 80-80, DP 60-60
- 1.6.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.6.12. RIE for dry oxide (LAM9400): 90 s etch time, 150 nm target etch depth, mnf_oxide1 (Recipe)
- 1.6.13. RIE for Si₃N₄ (LAM9400): 60 s etch time, 100 nm target etch depth, mnf_oxide1 (Recipe)
- 1.6.14. RIE for a-Si (LAM9400): 60 s etch time, 150 nm target etch depth, mnf_oxide1 (Recipe)
- 1.6.15. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.6.16. DI water Spin / Rinse / Dry
- 1.6.17. Sample cleaning (Yes plasma stripper): 800 W, 120 s, 150 °C, 80 sccm O₂

1.7. Metal openings - Bottom

- 1.7.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.7.2. DI water Spin / Rinse / Dry
- 1.7.3. Dehydration bake: 150 °C, 10 min
- 1.7.4. HMDS deposition (Yes image reversal oven): 300 s
- 1.7.5. PR coating (CEE spinner): SPR 7.0, 1000 rpm, 15 µm thick
- 1.7.6. Soft bake (Hot plate): 85 °C, 600 s

- 1.7.7. PR exposure (MA6): 80 s exposure time, hard contact, 30 µm gap
- 1.7.8. Post exposure delay: 4 hours to finish the photochemical reaction
- 1.7.9. Post exposure bake (Hot plate): 85 °C, 600 s
- 1.7.10. PR development (CEE developer): AZ726, DP 80-80, DP 80-80
- 1.7.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.7.12. RIE for dry oxide (LAM9400): 180 s etch time, 400 nm target etch depth, mnf_oxide1 (Recipe)
- 1.7.13. RIE for Si₃N₄ (LAM9400): 180 s etch time, 300 nm target etch depth, mnf_oxide1 (Recipe)
- 1.7.14. RIE for a-Si (LAM9400): 180 s etch time, 200 nm target etch depth, mnf_oxide1 (Recipe)
- 1.7.15. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.7.16. DI water Spin / Rinse / Dry
- 1.7.17. Sample cleaning (Yes plasma stripper): 800 W, 120 s, 150 °C, 80 sccm O₂

1.8. Metal patterning for top and bottom contacts

- 1.8.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.8.2. DI water Spin / Rinse / Dry
- 1.8.3. Pre-metal cleaning: 100:1 HF, 120 s etch time
- 1.8.4. Metal deposition (Enerjet): Aluminum, 300 nm target thickness
- 1.8.5. HMDS deposition (Yes image reversal oven): 300 s
- 1.8.6. PR coating (CEE spinner): SPR 3.0, 1000 rpm, 5 µm thick
- 1.8.7. Soft bake (Hot plate): 115 °C, 90 s
- 1.8.8. PR exposure (MA6): 24 s exposure time, hard contact, 30 μ m gap

- 1.8.9. Post exposure delay: 1-3 hours to finish the photochemical reaction
- 1.8.10. Post exposure bake (Hot plate): 115 °C, 90 s
- 1.8.11. Image reversal (Yes image reversal oven): 45 min
- 1.8.12. Flood exposure (MA6): 50 s exposure time, hard contact, 30 µm gap
- 1.8.13. Post exposure delay: 1-3 hours to finish the photochemical reaction
- 1.8.14. Post exposure bake (Hot plate): 115 °C, 90 s
- 1.8.15. PR development (CEE developer): AZ726, DP 80-80
- 1.8.16. Flood exposure (MA6): 30 s exposure time, hard contact, 30 µm gap
- 1.8.17. PR development (CEE developer): AZ726, DP 80-80
- 1.8.18. Hard baking (Hot plate): 110 °C, 300 s
- 1.8.19. Aluminum etching (Acid 73): 100 Å/s etch rate, 50 °C,

60 s etch time (10 s over etch)

- 1.8.20. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.8.21. DI water Spin / Rinse / Dry

1.9. Anti-reflection coating layer deposition

1.9.1. PECVD Si₃N₄ deposition for a-Si (GSI PECVD): 100 nm target thickness,

380 °C deposition temperature, low reflectance at $\lambda = 850$ nm

1.9.2. PECVD Si_3N_4 deposition for Si_3N_4 (GSI PECVD): 50 nm target thickness,

380 °C deposition temperature, low reflectance at $\lambda = 850$ nm

1.9.3. PECVD SiO₂ deposition for dry oxide (GSI PECVD): 100 nm target,

380 °C deposition temperature, low reflectance at $\lambda = 850$ nm

1.10. Via – Top

1.10.1. Wafer cleaning: 5 min Acetone, 5 min IPA

- 1.10.2. DI water Spin / Rinse / Dry
- 1.10.3. Dehydration bake: 150 °C, 10 min
- 1.10.4. HMDS deposition (Yes image reversal oven): 300 s
- 1.10.5. PR coating (CEE spinner): SPR 7.0, 1000 rpm, 15 µm thick
- 1.10.6. Soft bake (Hot plate): 85 °C, 600 s
- 1.10.7. PR exposure (MA6): 45 s exposure time, hard contact, 30 µm gap
- 1.10.8. Post exposure delay: 4 hours to finish the photochemical reaction
- 1.10.9. Post exposure bake (Hot plate): 85 °C, 600 s
- 1.10.10. PR development (CEE developer): AZ726, DP 80-80, DP 60-60
- 1.10.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.10.12. RIE for dry oxide (LAM9400): 90 s etch time, 120 nm target etch depth, mnf_oxide1 (Recipe)
- 1.10.13. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.10.14. DI water Spin / Rinse / Dry
- 1.10.15. Sample cleaning (Yes plasma stripper): 100 W, 60 s, 60 °C, 35 sccm O₂

1.11. Via – Bottom

- 1.11.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.11.2. DI water Spin / Rinse / Dry
- 1.11.3. Dehydration bake: 150 °C, 10 min
- 1.11.4. HMDS deposition (Yes image reversal oven): 300 s
- 1.11.5. PR coating (CEE spinner): SPR 7.0, 1000 rpm, 15 µm thick
- 1.11.6. Soft bake (Hot plate): 85 °C, 600 s
- 1.11.7. PR exposure (MA6): 80 s exposure time, hard contact, 30 µm gap

- 1.11.8. Post exposure delay: 4 hours to finish the photochemical reaction
- 1.11.9. Post exposure bake (Hot plate): 85 °C, 600 s
- 1.11.10. PR development (CEE developer): AZ726, DP 80-80 x2, DP 60-60
- 1.11.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.11.12. RIE (LAM9400): 180 s etch time, 270 nm target etch depth,

mnf_oxide1 (Recipe)

1.11.13. Wafer cleaning: 5 min Acetone, 5 min IPA

1.11.14. DI water Spin / Rinse / Dry

1.11.15. Sample cleaning (Yes plasma stripper): 100 W, 60 s, 60 °C, 35 sccm O₂

1.12. Bonding pad metallization

- 1.12.1. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.12.2. DI water Spin / Rinse / Dry
- 1.12.3. Aluminum oxide removal: 5 s etch time (Aluminum etchant type A),

30 Å/s etch rate

- 1.12.4. Aluminum deposition (Enerjet): 1 µm target thickness
- 1.12.5. HMDS deposition (Yes image reversal oven): 300 s
- 1.12.6. PR coating (CEE spinner): SPR 7.0, 2000 rpm, 10 µm thick
- 1.12.7. Soft bake (Hot plate): 85 °C, 330 s
- 1.12.8. PR exposure (MA6): 30 s exposure time, hard contact, 30 µm gap
- 1.12.9. Post exposure delay: 3 hours to finish the photochemical reaction
- 1.12.10. Post exposure bake (Hot plate): 85 °C, 90 s
- 1.12.11. PR development (CEE developer): AZ726, DP 80-80 x2
- 1.12.12. PR exposure (MA6): 20 s exposure time, hard contact, 30 µm gap

- 1.12.13. Post exposure bake (Hot plate): 85 °C, 60 s
- 1.12.14. PR development (CEE developer): AZ726, DP 80-80, DP 60-60
- 1.12.15. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 1.12.16. Hard baking (Hot plate): 110 °C, 300 s
- 1.12.17. Aluminum etching (Acid 73): 100 Å/s etch rate, 50 °C,

120 s etch time (20 s over etch)

- 1.12.18. Wafer cleaning: 5 min Acetone, 5 min IPA
- 1.12.19. DI water Spin / Rinse / Dry

2. GaAs Photovoltaic cells

2.1. Top contact for p-GaAs

- 2.1.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.1.2. HMDS (Yes plasma stripper): 300 s
- 2.1.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,
- 2.1.4. Soft bake (Hot plate): 115 °C, 90 s
- 2.1.5. PR exposure (MJB3): 9 s exposure time, hard contact
- 2.1.6. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.1.7. Image reversal (Yes plasma stripper): 45 min
- 2.1.8. Flood exposure (MJB3): 12 s exposure time, hard contact
- 2.1.9. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.1.10. PR development (CEE developer): AZ726, DP 40-40
- 2.1.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 2.1.12. Native oxide etch: NH₄OH: DI water = 10 ml : 200 ml, 30 s etch time
- 2.1.13. Metal deposition (Enerjet): Ti / Pt / Au = 50/100/200 nm
- 2.1.14. Metal lift-off: 10 min Acetone, 5 min IPA, 5 min DI water

2.2. Selective etch for heavily doped top p-GaAs

- 2.2.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.2.2. HMDS (Yes plasma stripper): 300 s
- 2.2.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,
- 2.2.4. Soft bake (Hot plate): 115 °C, 90 s
- 2.2.5. PR exposure (MJB3): 9 s exposure time, hard contact

- 2.2.6. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.2.7. PR development (CEE developer): AZ726, DP 40-40

2.2.8. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂

2.2.9. GaAs etch: Citric acid : H_2O_2 : DI water = 30 ml : 10 ml : 1500 ml,

8-10 min etch time, Surface color check (Dark blue)

2.2.10. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water

2.3. Device isolation down to n+ layer

- 2.3.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.3.2. HMDS (Yes plasma stripper): 300 s
- 2.3.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,
- 2.3.4. Soft bake (Hot plate): 115 °C, 90 s
- 2.3.5. PR exposure (MJB3): 9 s exposure time, hard contact
- 2.3.6. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.3.7. PR development (CEE developer): AZ726, DP 40-40
- 2.3.8. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 2.3.9. RIE (LAM9400): Cl_2 : BCl_3 : Ar = 20: 20 sccm, 4 μ m target depth

220 - 270 Å / s etch rate according to exposed area, 180 s etch time

2.3.10. Surface treatment : H_3PO_4 : H_2O_2 : DI water = 60 : 20 : 1000 ml,

10 s etch time

- 2.3.11. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.3.12. Plasma cleaning (Yes plasma stripper): 800 W, 120 s, 150 °C, 80 sccm O₂

2.4. Bottom contact metallization – n-GaAs

2.4.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water

2.4.2. HMDS (Yes plasma stripper): 300 s

2.4.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,

2.4.4. Soft bake (Hot plate): 115 °C, 90 s

2.4.5. PR exposure (MJB3): 9 s exposure time, hard contact

2.4.6. Post exposure bake (Hot plate): 115 °C, 90 s

2.4.7. Image reversal (Yes plasma stripper): 45 min

2.4.8. Flood exposure (MJB3): 16 s exposure time, hard contact

2.4.9. Post exposure bake (Hot plate): 115 °C, 90 s

2.4.10. PR development (CEE developer): AZ726, DP 40-40

2.4.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂

2.4.12. Native oxide etch: NH_4OH : DI water = 10 ml : 200 ml, 30 s etch time

2.4.13. Metal deposition (Enerjet): Ni/Ge/Au/Ti/Pt/Au = 5/30/100/50/100/200 nm

2.4.14. Metal lift-off: 10 min Acetone, 5 min IPA, 5 min DI water

2.5. Device isolation down to n+ layer

2.5.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water

2.5.2. HMDS (Yes plasma stripper): 300 s

2.5.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,

2.5.4. Soft bake (Hot plate): 115 °C, 90 s

2.5.5. PR exposure (MJB3): 14 s exposure time, hard contact

2.5.6. Post exposure bake (Hot plate): 115 °C, 90 s

2.5.7. PR development (CEE developer): AZ726, DP 40-40

2.5.8. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂

2.5.9. RIE (LAM9400): Cl_2 : BCl₃: Ar = 20 : 20 sccm, 5.7 µm target depth

220 - 270 Å / s etch rate according to exposed area, 180 s etch time

2.5.10. Surface treatment : H_3PO_4 : H_2O_2 : DI water = 60 : 20 : 1000 ml,

10 s etch time

- 2.5.11. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.5.12. Plasma cleaning (Yes plasma stripper): 800 W, 120 s, 150 °C, 80 sccm O₂

2.6. Ammonium sulfide ((NH4)₂S) treatment for sidewall/surface passivation

2.6.1. Ammonium sulfide ((NH4)₂S): 23 %, 10 min treatment

2.6.2. DI water clean

2.7. PECVD Silicon nitride deposition for AR coating and passivation

2.7.1. PECVD Si₃N₄ deposition (GSI PECVD): 100 nm target thickness,

380 °C deposition temperature, low reflectance at $\lambda = 850$ nm

2.8. Via – Top and bottom pads

- 2.8.1. Wafer cleaning: 5 min Acetone, 5 min IPA. 5 min DI water
- 2.8.2. HMDS deposition (Yes image reversal oven): 300 s
- 2.8.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,
- 2.8.4. Soft bake (Hot plate): 115 °C, 90 s
- 2.8.5. PR exposure (MJB3): 14 s exposure time, hard contact
- 2.8.6. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.8.7. PR development (CEE developer): AZ726, DP 40-40
- 2.8.8. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 2.8.9. RIE (LAM9400): mnf_oxide1 (Recipe), 120 s etch time, 180 nm target
- 2.8.10. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.8.11. Plasma cleaning (Yes plasma stripper): 800 W, 120 s, 150 °C, 80 sccm O₂

2.9. Bonding pad metallization

- 2.9.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.9.2. HMDS (Yes plasma stripper): 300 s
- 2.9.3. PR coating (CEE spinner): SPR 3.0, 4000 rpm, 2.2 um thick,
- 2.9.4. Soft bake (Hot plate): 115 °C, 90 s
- 2.9.5. PR exposure (MJB3): 14 s exposure time, hard contact
- 2.9.6. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.9.7. Image reversal (Yes plasma stripper): 45 min
- 2.9.8. Flood exposure (MJB3): 16 s exposure time, hard contact
- 2.9.9. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.9.10. PR development (CEE developer): AZ726, DP 40-40
- 2.9.11. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂
- 2.9.12. Metal deposition (Enerjet): Ti/Al/Ti/Au = 10/7000/10/300 nm
- 2.9.13. Metal lift-off: 10 min Acetone, 5 min IPA, 5 min DI water

2.10. Bottom contact metallization – n-GaAs

- 2.10.1. Wafer cleaning: 5 min Acetone, 5 min IPA, 5 min DI water
- 2.10.2. Dehydration bake (Hot plate): 115 °C, 60 s
- 2.10.3. PR coating (CEE spinner): LOR 10B, 4000 rpm, 0.9 um thick,
- 2.10.4. Soft bake (Hot plate): 165 °C, 180 s
- 2.10.5. PR exposure (MJB3): 14 s exposure time, hard contact
- 2.10.6. Post exposure bake (Hot plate): 115 °C, 90 s
- 2.10.7. PR development (CEE developer): AZ726, DP 40-40
- 2.10.8. Plasma ashing (Yes plasma stripper): 100 W, 60 s, 35 sccm O₂

2.10.9. Metal deposition (Enerjet): Ti/Au = 10/500 nm

2.10.10. Metal lift-off: 20 min Remover PG at 65 °C, 10 min IPA, 5 min DI water

Bibliography

- Oh, Sechang, et al. "IoT 2—the Internet of Tiny Things: Realizing mm-Scale Sensors through 3D Die Stacking." 2019 Design, Automation & Test in Europe Conference & Exhibition (DATE). IEEE, 2019.
- Lee, Inhee, et al. "A 10mm 3 Light-Dose Sensing IoT 2 System with 35-to-339nW 10-to-300klx Light-Dose-to-Digital Converter." 2019 Symposium on VLSI Technology. IEEE, 2019.
- Cadei, Andrea, et al. "Kinetic and thermal energy harvesters for implantable medical devices and biomedical autonomous sensors." Measurement Science and Technology 25.1 (2013): 012003.
- Platt, Stephen R., et al. "The use of piezoelectric ceramics for electric power generation within orthopedic implants." IEEE/ASME transactions on mechatronics 10.4 (2005): 455-461.
- Kim, Je-Dok, et al. "A proposal on wireless power transfer for medical implantable applications based on reviews." 2014 IEEE Wireless Power Transfer Conference. IEEE, 2014.
- Lee, Seung Bae, et al. "An inductively powered scalable 32-channel wireless neural recording system-on-a-chip for neuroscience applications." IEEE transactions on biomedical circuits and systems 4.6 (2010): 360-371.
- S. Roundy, P. K. Wright, and J. M. Rabaey, Energy Scavenging for Wireless Sensor Networks, Boston, MA, Springer, 2004.

- Lee, Yoonmyung, et al. "A modular 1 mm3 die-stacked sensing platform with low power I2C inter-die communication and multi-modal energy harvesting." IEEE Journal of Solid-State Circuits 48.1 (2012): 229-243.
- Jung, Wanyeong, et al. "An ultra-low power fully integrated energy harvester based on self-oscillating switched-capacitor voltage doubler." IEEE Journal of Solid-State Circuits 49.12 (2014): 2800-2811.
- 10. Hanson, Scott, et al. "A low-voltage processor for sensing applications with picowatt standby mode." IEEE Journal of Solid-State Circuits 44.4 (2009): 1145-1155.
- 11. Fojtik, Matthew, et al. "A millimeter-scale energy-autonomous sensor system with stacked battery and solar cells." IEEE Journal of Solid-State Circuits 48.3 (2013): 801-813.
- 12. Riordan, C., and R. Hulstron. "What is an air mass 1.5 spectrum? (Solar cell performance calculations)." IEEE Conference on Photovoltaic Specialists. IEEE, 1990.
- 13. Teran, Alan S., Eunseong Moon, Wootaek Lim, Gyouho Kim, Inhee Lee, David Blaauw, and Jamie D. Phillips. "Energy harvesting for GaAs photovoltaics under low-flux indoor lighting conditions." IEEE transactions on electron devices 63, no. 7 (2016): 2820-2825.
- 14. Moon, Eunseong, David Blaauw, and Jamie D. Phillips. "Infrared Energy Harvesting in Millimeter-Scale GaAs Photovoltaics." IEEE transactions on electron devices 64, no. 11 (2017): 4554-4560.
- 15. Freunek, Monika, Michael Freunek, and Leonhard M. Reindl. "Maximum efficiencies of indoor photovoltaic devices." IEEE Journal of Photovoltaics 3, no. 1 (2013): 59-64.
- Mathews, Ian, Paul J. King, Frank Stafford, and Ronan Frizzell. "Performance of III–V solar cells as indoor light energy harvesters." IEEE Journal of Photovoltaics 6, no. 1 (2016): 230-235.

- Martí Vega, Antonio, and Gerardo Lopez Araujo. "Limiting efficiencies of GaAs solar cells." IEEE Transactions on Electron Devices 37, no. 5 (1990): 1402-1405.
- Kimovec R, Helmers H, Bett AW, Topič M. Comprehensive electrical loss analysis of monolithic interconnected multi-segment laser power converters. Prog Photovolt Res Appl. 2018;1–11.
- 19. Moon, Eunseong, David Blaauw, and Jamie D. Phillips. "Subcutaneous photovoltaic infrared energy harvesting for bio-implantable devices." IEEE transactions on electron devices 64.5 (2017): 2432-2437.
- 20. Smith, Andrew M., Michael C. Mancini, and Shuming Nie. "Bioimaging: second window for in vivo imaging." Nature nanotechnology 4.11 (2009): 710.
- 21. Teran, Alan S., et al. "AlGaAs photovoltaics for indoor energy harvesting in mm-scale wireless sensor nodes." IEEE Transactions on Electron Devices 62.7 (2015): 2170-2175.
- 22. Murakawa, Koji, Makoto Kobayashi, Osamu Nakamura, and Satoshi Kawata. "A wireless near-infrared energy system for medical implants." IEEE Engineering in Medicine and Biology Magazine 18, no. 6 (1999): 70-72.
- 23. Chung, Hoon, Tianhong Dai, Sulbha K. Sharma, Ying-Ying Huang, James D. Carroll, and Michael R. Hamblin. "The nuts and bolts of low-level laser (light) therapy." Annals of biomedical engineering 40, no. 2 (2012): 516-533.
- 24. Ayazian, Sahar, et al. "A photovoltaic-driven and energy-autonomous CMOS implantable sensor." IEEE Transactions on biomedical circuits and systems 6.4 (2012): 336-343.
- 25. Bunea, Gabriela E., Karen E. Wilson, Yevgeny Meydbray, Matthew P. Campbell, and Denis M. De Ceuster. "Low light performance of mono-crystalline silicon solar cells." In 2006 IEEE 4th World Conference on Photovoltaic Energy Conference, vol. 2, pp. 1312-

1314. IEEE, 2006.

- 26. Reich, N. H., W. G. J. H. M. Van Sark, E. A. Alsema, R. W. Lof, R. E. I. Schropp, W. C. Sinke, and W. C. Turkenburg. "Crystalline silicon cell performance at low light intensities." Solar Energy Materials and Solar Cells 93, no. 9 (2009): 1471-1481.
- 27. Desa, MK Mat, et al. "Silicon back contact solar cell configuration: A pathway towards higher efficiency." Renewable and sustainable energy reviews 60 (2016): 1516-1532.
- Green, Martin A., et al. "Solar cell efficiency tables (version 54)." Progress in Photovoltaics: Research and Applications 27.7 (2019): 565-575.
- 29. Zhao, Jianhua, et al. "24% efficient perl silicon solar cell: recent improvements in high efficiency silicon cell research." Solar energy materials and solar cells 41 (1996): 87-99.
- Guide, Sentaurus Device User, and E. Version. "Synopsys Inc." Mountain View, CA, USA (2013).
- Green, Martin A., and Mark J. Keevers. "Optical properties of intrinsic silicon at 300 K." Progress in Photovoltaics: Research and Applications 3.3 (1995): 189-192.
- 32. Ren, Yongling, et al. "Charge stability in LPCVD silicon nitride for surface passivation of silicon solar cells." 2010 35th IEEE Photovoltaic Specialists Conference. IEEE, 2010.
- 33. Rühle, K., M. Rauer, M. Rüdiger, J. Giesecke, T. Niewelt, C. Schmiga, S. W. Glunz, and M. Kasemann. "Passivation layers for indoor solar cells at low irradiation intensities." Energy Procedia 27 (2012): 406-411.
- 34. Hoex, Bram, Jan Schmidt, P. Pohl, M. C. M. Van de Sanden, and W. M. M. Kessels.
 "Silicon surface passivation by atomic layer deposited Al2O3." Journal of Applied Physics 104, no. 4 (2008): 044903
- 35. Angermann, H., J. Rappich, L. Korte, I. Sieber, E. Conrad, M. Schmidt, K. Hübener, J.

Polte, and J. Hauschild. "Wet-chemical passivation of atomically flat and structured silicon substrates for solar cell application." Applied Surface Science 254, no. 12 (2008): 3615-3625.

- 36. Angermann, H., W. Henrion, M. Rebien, and A. Röseler. "Wet-chemical passivation and characterization of silicon interfaces for solar cell applications." Solar energy materials and solar cells 83, no. 4 (2004): 331-346.
- 37. Song, G., M. Y. Ali, and Meng Tao. "A high Schottky-barrier of 1.1 eV between Al and S-passivated p-type Si (100) surface." IEEE electron device letters 28.1 (2007): 71-73.
- 38. Tao, Meng, and Muhammad Yusuf Ali. "Effect of sulfur passivation of silicon "100... on Schottky barrier height: Surface states versus surface dipole." (2007).
- 39. Zhang, Hai-feng, Arunodoy Saha, Wen-cheng Sun, and Meng Tao. "Characterization of Al/Si junctions on Si (100) wafers with chemical vapor deposition-based sulfur passivation." Applied Physics A 116, no. 4 (2014): 2031-2038.
- 40. Saha, Arunodoy, Haifeng Zhang, Wen-Cheng Sun, and Meng Tao. "Grain boundary passivation in multicrystalline silicon using hydrogen sulfide." ECS Journal of Solid State Science and Technology 4, no. 7 (2015): P186-P189.
- 41. Shockley, William, and Hans J. Queisser. "Detailed balance limit of efficiency of p-n junction solar cells." Journal of applied physics 32.3 (1961): 510-519.
- 42. IXYS Corporation: KXOB22-12X1 (2010) https://docs.google.com/view erng/viewer?url=http://ixapps.ixys.com/DataSheet/KXOB22-12X1L-DATA-SHEET-20130902-.pdf
- 43. Haeberlin, Andreas, Adrian Zurbuchen, Sébastien Walpen, Jakob Schaerer, Thomas Niederhauser, Christoph Huber, Hildegard Tanner et al. "The first batteryless, solar-

powered cardiac pacemaker." Heart rhythm 12, no. 6 (2015): 1317-1323.

- 44. Song, Kwangsun, Jung Hyun Han, Taehoon Lim, Namyun Kim, Sungho Shin, Juho Kim, Hyuck Choo et al. "Subdermal flexible solar cell arrays for powering medical electronic implants." Advanced healthcare materials 5, no. 13 (2016): 1572-1580.
- 45. Goto, Kazuya, Tetsuya Nakagawa, Osamu Nakamura, and Satoshi Kawata. "An implantable power supply with an optically rechargeable lithium battery." IEEE Transactions on Biomedical Engineering 48, no. 7 (2001): 830-833.
- 46. Kayes, Brendan M., Hui Nie, Rose Twist, Sylvia G. Spruytte, Frank Reinhardt, Isik C. Kizilyalli, and Gregg S. Higashi. "27.6% conversion efficiency, a new record for single-junction solar cells under 1 sun illumination." In Photovoltaic Specialists Conference (PVSC), 2011 37th IEEE, pp. 000004-000008. IEEE, 2011.
- 47. Reich, N. H., W. G. J. H. M. Van Sark, E. A. Alsema, R. W. Lof, R. E. I. Schropp, W. C. Sinke, and W. C. Turkenburg. "Crystalline silicon cell performance at low light intensities." Solar Energy Materials and Solar Cells 93, no. 9 (2009): 1471-1481.
- 48. Hannan, Mahammad A., Saad Mutashar, Salina A. Samad, and Aini Hussain. "Energy harvesting for the implantable biomedical devices: issues and challenges." Biomedical engineering online 13, no. 1 (2014): 79.
- 49. Remashan, K., and K. N. Bhat. "Combined effect of (NH4) 2Sx treatment and postmetallization annealing with plasma-enhanced chemical vapour deposition silicon nitride gate dielectric on the GaAs metal-insulator-semiconductor characteristics and the photoluminescence characteristics of GaAs substrates." Semiconductor science and technology 17, no. 3 (2002): 243.
- 50. Oigawa, Haruhiro, Jia-Fa Fan, Yasuo Nannichi, Hirohiko Sugahara, and Masaharu

Oshima. "Universal passivation effect of (NH4) 2Sx treatment on the surface of III-V compound semiconductors." Japanese Journal of Applied Physics 30, no. 3A (1991): L322.

- 51. Carpenter, M. S., M. R. Melloch, M. S. Lundstrom, and S. P. Tobin. "Effects of Na2S and (NH4) 2S edge passivation treatments on the dark current-voltage characteristics of GaAs pn diodes." Applied physics letters 52, no. 25 (1988): 2157-2159.
- 52. Xuan, Yi, Hung-Chun Lin, and D. Ye Peide. "Simplified Surface Preparation for GaAs Passivation Using Atomic Layer-Deposited High-k Dielectrics." IEEE Transactions on Electron Devices 54, no. 8 (2007): 1811-1817.
- 53. Remashan, K., and K. N. Bhat. "Silicon nitride/(NH4)2Sx passivation of n-GaAs to unpin the Fermi level." Electronics Letters 32, no. 7 (1996): 694-695.
- 54. Sheldon, Matthew T., Carissa N. Eisler, and Harry A. Atwater. "GaAs passivation with trioctylphosphine sulfide for enhanced solar cell efficiency and durability." Advanced Energy Materials 2, no. 3 (2012): 339-344.
- 55. Eisler, Carissa N., Matthew T. Sheldon, and Harry A. Atwater. "Enhanced performance of small GaAs solar cells via edge and surface passivation with trioctylphosphine sulfide." In Photovoltaic Specialists Conference (PVSC), 2012 38th IEEE, pp. 000937-000940. IEEE, 2012.
- 56. Stellwag, T. B., Mr R. Melloch, M. S. Lundstrom, M. S. Carpenter, and R. F. Pierret.
 "Orientation-dependent perimeter recombination in GaAs diodes." Applied physics letters
 56, no. 17 (1990): 1658-1660.
- 57. Henry, C. H., R. A. Logan, and F. R. Merritt. "The effect of surface recombination on current in Al x Ga1– x As heterojunctions." Journal of Applied Physics 49, no. 6 (1978):

3530-3542.

- 58. Mariani, Giacomo, Adam C. Scofield, Chung-Hong Hung, and Diana L. Huffaker. "GaAs nanopillar-array solar cells employing in situ surface passivation." Nature communications 4 (2013): 1497.
- 59. Chang, Chia-Chi, Chun-Yung Chi, Maoqing Yao, Ningfeng Huang, Chun-Chung Chen, Jesse Theiss, Adam W. Bushmaker et al. "Electrical and optical characterization of surface passivation in GaAs nanowires." Nano letters 12, no. 9 (2012): 4484-4489.
- 60. Mariani, Giacomo, Ping-Show Wong, Aaron M. Katzenmeyer, Francois Léonard, Joshua Shapiro, and Diana L. Huffaker. "Patterned radial GaAs nanopillar solar cells." Nano letters 11, no. 6 (2011): 2490-2494.
- M. C. Frost and M. E. Meyerhoff, "Implantable chemical sensors for real-time clinical monitoring: Progress and challenges," Current Opinion Chem. Biol., vol. 6, no. 5, pp. 633– 641, 2002.
- R. K. Meruva and M. E. Meyerhoff, "Catheter-type sensor for potentiometric monitoring of oxygen, pH and carbon dioxide," Biosens.Bioelectron., vol. 13, no. 2, pp. 201–212, 1998.
- 63. D. S. Bindra et al., "Design and in vitro studies of a needle-type glucose sensor for subcutaneous monitoring," Anal. Chem., vol. 63, no. 17, pp. 1692–1696, 1991.
- 64. X. Chen, N. Matsumoto, Y. Hu, and G. S. Wilson, "Electrochemically mediated electrodeposition/electropolymerization to yield a glucose microbiosensor with improved characteristics," Anal. Chem., vol. 74, no. 2, pp. 368–372, 2002.
- 65. S. Oh et al., "A dual-slope capacitance-to-digital converter integrated in an implantable pressure-sensing system," IEEE J. Solid-State Circuits, vol. 50, no. 7, pp. 1581–1591, Jul.

2015.

- 66. S. Jeong, Z. Foo. Y. Lee, J.-Y. Sim, D. Blaauw, and D. Sylvester, "A fully-integrated 71 nW CMOS temperature sensor for low power wireless sensor nodes," IEEE J. Solid-State Circuits, vol. 49, no. 8, pp. 1682–1693, Aug. 2014.
- 67. G. Kim et al., "A millimeter-scale wireless imaging system with continuous motion detection and energy harvesting," in Symp. VLSI Circuits Dig. Tech. Papers, Jun. 2014, pp. 1–2.
- Y. Lee, D. Blaauw, and D. Sylvester, "Ultralow power circuit design for wireless sensor nodes for structural health monitoring," Proc. IEEE, vol. 104, no. 8, pp. 1529–1546, Aug. 2016.
- R. Venkatasubramanian, E. Siivola, T. Colpitts, and B. O'Quinn, "Thin-film thermoelectric devices with high room-temperature figures of merit," Nature, vol. 413, no. 6856, pp. 597–602, 2001.
- 70. Y. Yang, X.-J. Wei, and J. Liu, "Suitability of a thermoelectric power generator for implantable medical electronic devices," J. Phys. D, Appl. Phys., vol. 40, no. 18, p. 5790, 2007.
- 71. S. Almouahed, M. Gouriou, C. Hamitouche, E. Stindel, and C. Roux, "Self-powered instrumented knee implant for early detection of postoperative complications," in Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC), Aug. 2010, pp. 5121–5124.
- 72. B. E. Lewandowski, K. L. Kilgore, and K. J. Gustafson, "Design considerations for an implantable, muscle powered piezoelectric system for generating electrical power," Ann. Biomed. Eng., vol. 35, no. 4, pp. 631–641, 2007.
- 73. R. Bashirullah, "Wireless implants," IEEE Microw. Mag., vol. 11, no. 7, pp. S14-S23,

Dec. 2010.

- 74. R.-F. Xue, K.-W. Cheng, and M. Je, "High-efficiency wireless power transfer for biomedical implants by optimal resonant load transformation," IEEE Trans. Circuits Syst. I, Reg. Papers, vol. 60, no. 4, pp. 867–874, Apr. 2013.
- 75. A. K. RamRakhyani, S. Mirabbasi, and M. Chiao, "Design and optimization of resonancebased efficient wireless power delivery systems for biomedical implants," IEEE Trans. Biomed. Circuits Syst., vol. 5, no. 1, pp. 48–63, Feb. 2011.
- 76. M. Catrysse, B. Hermans, and R. Puers, "An inductive power system with integrated bidirectional data-transmission," Sens. Actuators A, Phys., vol. 115, nos. 2–3, pp. 221–229, Sep. 2004.
- 77. A. N. Bashkatov, E. A. Genina, V. I. Kochubey, and V. V. Tuchin, "Optical properties of human skin, subcutaneous and mucous tissues in the wavelength range from 400 to 2000 nm," J. Phys. D, Appl. Phys., vol. 38, no. 15, p. 2543, 2005.
- 78. L. A. Sordillo, Y. Pu, S. Pratavieira, Y. Budansky, and R. R. Alfano, "Deep optical imaging of tissue using the second and third near-infrared spectral windows," J. Biomed. Opt., vol. 19, no. 5, p. 056004, 2014.
- 79. Jacques, Steven L. "Optical properties of biological tissues: a review." Physics in Medicine & Biology 58.11 (2013): R37.
- 80. Genina, E. A., et al. "Optical clearing of human dura mater." Optics and Spectroscopy 98.3 (2005): 470-476.
- 81. Nomura, Yasutomo, Biomedical Science and Engineering, 2015
- 82. LIo, America. "American National Standard for Safe Use of Lasers (ANSI Z136. 1)." Laser Institute of America, Orlando, Florida (2014).

- 83. Wu, Xiao, et al. "A 0.04 MM 3 16NW Wireless and Batteryless Sensor System with Integrated Cortex-M0+ Processor and Optical Communication for Cellular Temperature Measurement." 2018 IEEE Symposium on VLSI Circuits. IEEE, 2018.
- 84. Wolf PD (BMI) In: Reichert WM, editor. Indwelling Neural Implants: Strategies for Conending with the In Vivo Environment. Boca Raton (FL): CRC Press/Taylor & Francis; 2008. Chapter 3. Available from: https://www.ncbi.nlm.nih.gov/books/NBK3932/
- 85. J. C. C. Fan, "Theoretical temperature dependence of solar cell parameters," Solar cells, vol. 17, nos. 2–3, pp. 309–315, 1986.
- 86. N. J. Vardaxis, T. A. Brans, M. E. Boon, R. W. Kreis, and L. M. Marres, "Confocal laser scanning microscopy of porcine skin: Implications for human wound healing studies," J. Anatomy, vol. 190, no. 4, pp. 601–611, 1997.
- 87. E. Zamora-Rojas, B. Aernouts, A. Garrido-Varo, D. Pérez-Marín, J. E. Guerrero-Ginel, and W. Saeys, "Double integrating sphere measurements for estimating optical properties of pig subcutaneous adipose tissue," Innov. Food Sci. Emerg. Technol., vol. 19, pp. 218– 226, Jul. 2013.
- 88. M. R. Arnfield, R. P. Mathew, J. Tulip, and M. S. McPhee, "Analysis of tissue optical coefficients using an approximate equation valid for comparable absorption and scattering," Phys. Med. Biol., vol. 37, no. 6, p. 1219, 1992.
- 89. Y. Du, X. H. Hu, M. Cariveau, X. Ma, G. W. Kalmus, and J. Q. Lu, "Optical properties of porcine skin dermis between 900 nm and 1500 nm," Phys. Med. Biol., vol. 46, no. 1, p. 167, 2001.
- 90. W. Meyer, R. Schwarz, and K. Neurand, "The skin of domestic mammals as a model for the human skin, with special reference to the domestic pig1," in Skin-Drug Application

and Evaluation of Environmental Hazards. Basel, Switzerland: Karger Publishers, 1978, pp. 39–52.

- 91. I. S. Saidi, S. L. Jacques, and F. K. Tittel, "Mie and Rayleigh modeling of visible-light scattering in neonatal skin," Appl. Opt., vol. 34, no. 31, pp. 7410–7418, 1995.
- 92. I. Lee, W. Lim, A. Teran, J. Phillips, D. Sylvester, and D. Blaauw, "A >78%-efficient light harvester over 100-to-100klux with reconfigurable PV-cell network and MPPT circuit," in IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers, Jan. 2016, pp. 370– 371.
- 93. S. J. Kim et al., "The potential role of polymethyl methacrylate as a new packaging material for the implantable medical device in the bladder," BioMed Res. Int., vol. 2015, Feb. 2015, Art. no. 852456.
- 94. S. L. Peterson, A. McDonald, P. L. Gourley, and D. Y. Sasaki, "Poly(dimethylsiloxane) thin films as biocompatible coatings for microfluidic devices: Cell culture and flow studies with glial cells," J. Biomed. Mater. Res. A, vol. 72, no. 1, pp. 10–18, 2005.
- 95. J. H.-C. Chang, Y. Liu, and Y.-C. Tai, "Long term glass-encapsulated packaging for implant electronics," in Proc. IEEE 27th Int. Conf. Micro Elect. Mech. Syst. (MEMS), Jan. 2014, pp. 1127–1130.
- 96. Pena, Rafael, and Carlos Algora. "The influence of monolithic series connection on the efficiency of GaAs photovoltaic converters for monochromatic illumination." IEEE Transactions on Electron Devices 48, no. 2 (2001): 196-203.
- 97. Takeshiro, Y., Y. Okamoto, and Y. Mita. "Mask-programmable on-chip photovoltaic cell array." In Journal of Physics: Conference Series, vol. 1052, no. 1, p. 012144. IOP Publishing, 2018.

- 98. Lee, Inhee, Gyouho Kim, Eunseong Moon, Seokhyeon Jeong, Dongkwun Kim, Jamie Phillips, and David Blaauw. "A 179-Lux Energy-Autonomous Fully-Encapsulated 17-mm
 3 Sensor Node with Initial Charge Delay Circuit for Battery Protection." In 2018 IEEE Symposium on VLSI Circuits, pp. 251-252. IEEE, 2018.
- 99. Gai, Boju, Yukun Sun, Haneol Lim, Huandong Chen, Joseph Faucher, Minjoo L. Lee, and Jongseung Yoon. "Multilayer-Grown Ultrathin Nanostructured GaAs Solar Cells as a Cost-Competitive Materials Platform for III–V Photovoltaics." ACS nano 11, no. 1 (2017): 992-999
- 100. Datas, A., and P. G. Linares. "Monolithic interconnected modules (MIM) for high irradiance photovoltaic energy conversion: A comprehensive review." Renewable and Sustainable Energy Reviews 73 (2017): 477-495.
- 101. Sasaki, Kazuaki, et al. "Development of InGaP/GaAs/InGaAs inverted triple junction concentrator solar cells." AIP conference proceedings. Vol. 1556. No. 1. American Institute of Physics, 2013.
- 102. Geisz, J. F., et al. "40.8% efficient inverted triple-junction solar cell with two independently metamorphic junctions." Applied Physics Letters 93.12 (2008): 123505.
- 103. Tatavarti, Rao, et al. "InGaP/GaAs/InGaAs inverted metamorphic (IMM) solar cells on 4 "epitaxial lifted off (ELO) wafers." 2010 35th IEEE Photovoltaic Specialists Conference. IEEE, 2010.
- Essig, Stephanie, et al. "Raising the one-sun conversion efficiency of III–V/Si solar cells to 32.8% for two junctions and 35.9% for three junctions." Nature Energy 2.9 (2017):
 17144. Fig 1.
- 105. Fafard, Simon, Francine Proulx, Mark CA York, L. S. Richard, Philippe-Olivier

Provost, Richard Arès, Vincent Aimez, and Denis P. Masson. "High-photovoltage GaAs vertical epitaxial monolithic heterostructures with 20 thin p/n junctions and a conversion efficiency of 60%." Applied Physics Letters 109, no. 13 (2016): 131107. https://www.nrel.gov/pv/high-efficiency-iii-v-solar-cells.html

- 106. Lee, Kyusang, et al. "Non-destructive wafer recycling for low-cost thin-film flexible optoelectronics." Advanced Functional Materials 24.27 (2014): 4284-4291.
- 107. Normann, R.A., Maynard, E.M., Rousche, P.J. and Warren, D.J., 1999. A neural interface for a cortical vision prosthesis. Vision research, 39(15), pp.2577-2587.
- C.M. Lopez et al., "22.7 A 966-Electrode Neural Probe with 384 Configurable Channels in 0.13μm SOI CMOS," ISSCC, pp. 392-393, Feb. 2016.
- B.C. Johnson et al., "An Implantable 700μW 64-Channel Neuromodulation IC for Simultaneous Recording and Stimulation with Rapid Artifact Recovery," IEEE Symp. VLSI Circuits, pp. C48-C49, 2017.
- 110. H. Chandrakumar and D. Markovic, "27.1 A 2.8µW 80mVpp-Linear-InputRange
 1.6GΩ-Input Impedance Bio-Signal Chopper Amplifier Tolerant to Common-Mode
 Interference up to 650mVpp," ISSCC, pp. 448–449, Feb. 2017.
- 111. C. Kim et al., "A 92dB Dynamic Range Sub-µVrms-Noise 0.8µW/ch NeuralRecording ADC Array with Predictive Digital Autoranging," ISSCC, pp. 470–471, Feb. 2018.
- J. Lee et al., "An Implantable Wireless Network of Distributed Microscale Sensors for Neural Applications," Int. IEEE/EMBS NER, pp.871-874, 2019.
- III3. J. Lee et al., "Wireless Power and Data Link for Ensembles of Sub-mm scaleImplantable Sensors near 1GHz," IEEE BioCAS, pp.1-4, 2018.

- Lim, Jongyup, et al. "26.9 A 0.19× 0.17 mm 2 Wireless Neural Recording IC for Motor Prediction with Near-Infrared-Based Power and Data Telemetry." 2020 IEEE International Solid-State Circuits Conference-(ISSCC). IEEE, 2020.
- 115. M.M. Ghanbari et al., "A 0.8mm2 Ultrasonic Implantable Wireless Neural Recording System with Linear AM Backscattering," ISSCC, pp. 284-285, Feb. 2019.
- H.R. Eggert and V. Blazek, "Optical Properties of Human Brain Tissue, Meninges, and Brain Tumors in the Spectral Range of 200 to 900 nm", Neurosurgery, 21 (4), pp 459-464 (1987).
- 117. Lee, Sunwoo, et al. "A 330µm× 90µm opto-electronically integrated wireless system-on-chip for recording of neural activities." 2018 IEEE International Solid-State Circuits Conference-(ISSCC). IEEE, 2018.
- 118. Lee, Sunwoo, et al. "A 250 μm× 57 μm Microscale Opto-electronically Transduced Electrodes (MOTEs) for Neural Recording." IEEE transactions on biomedical circuits and systems 12.6 (2018): 1256-1266.
- 119. Kim, Tae-il, et al. "Injectable, cellular-scale optoelectronics with applications for wireless optogenetics." Science 340.6129 (2013): 211-216.
- 120. Pellegrini, Sara, et al. "Design and performance of an InGaAs-InP single-photon avalanche diode detector." IEEE journal of quantum electronics 42.4 (2006): 397-403.
- 121. Xu, Hesong, et al. "Design and characterization of a p+/n-well SPAD array in150nm CMOS process." Optics express 25.11 (2017): 12765-12778.
- Peters, Victor George, et al. "Optical properties of normal and diseased human breast tissues in the visible and near infrared." Physics in Medicine & Biology 35.9 (1990): 1317.

- 123. Elsherbini, A. A., S. M. Liff, and J. M. Swan. "Heterogeneous Integration Using Omni-Directional Interconnect Packaging." 2019 IEEE International Electron Devices Meeting (IEDM). IEEE, 2019.
- Ingerly, D. B., et al. "Foveros: 3D Integration and the use of Face-to-Face Chip Stacking for Logic Devices." 2019 IEEE International Electron Devices Meeting (IEDM).
 IEEE, 2019.
- R. Windisch et al., "40% efficient thin-film surface-textured light-emitting diodes by optimization of natural lithography," IEEE Transactions on Electronic Devices 47, 1492-148 (2000)
- 126. Vigneron, P. B., et al. "Advanced and reliable GaAs/AlGaAs ICP-DRIE etching for optoelectronic, microelectronic and microsystem applications." Microelectronic Engineering 202 (2018): 42-50.
- 127. Laermer, Franz, and Andrea Urban. "Challenges, developments and applications of silicon deep reactive ion etching." Microelectronic Engineering 67 (2003): 349-355.
- 128. Calandri, Niccolò, et al. "Optical crosstalk in InGaAs/InP SPAD array: Analysis and reduction with FIB-etched trenches." IEEE Photonics Technology Letters 28.16 (2016): 1767-1770.
- 129. Bérubé, Benoit-Louis, et al. "Development of a single photon avalanche diode (SPAD) array in high voltage CMOS 0.8 µm dedicated to a 3D integrated circuit (3DIC)."
 2012 IEEE Nuclear Science Symposium and Medical Imaging Conference Record (NSS/MIC). IEEE, 2012.
- Steranka, Frank M. "AlGaAs red light-emitting diodes." Semiconductors and Semimetals. Vol. 48. Elsevier, 1997. 65-96.

131. Yam, F. K., and Z. Hassan. "Innovative advances in LED technology." Microelectronics Journal 36.2 (2005): 129-137